Free for Children? Patient Cost-sharing and Health Care Utilization*

Toshiaki Iizuka[†]

Hitoshi Shigeoka[‡]

The University of Tokyo

Simon Fraser University and NBER

March, 2018

Abstract: Understanding how patient responds to price is a key to the optimal design of health insurance. However, past studies are predominantly concentrated on the adults and elderly, and surprisingly little is known about children. We examine the effect of patient cost-sharing on health care utilization among children by exploiting newly collected data on drastic subsidy expansion in Japan, with more than 5,000 changes in subsidy status at municipality-age-time level. We find that free care for children significantly increases spending on outpatient care by 22–31%, with the arcelasticity of -0.1 for all ages 7-14, which is smaller than the conventional estimate for adults. Interestingly, we find little evidence of asymmetric responses to the price changes of the opposite directions, implying that policy makers can reasonably employ existing elasticity estimates, regardless of the direction of the price changes. Finally, we provide suggestive evidence that that increases mostly reflect low-value or costly care. Increases in outpatient visits do not translate to reduction in hospitalization by "avoidable" conditions nor reduction in mortality. Furthermore, we document that inappropriate use of antibiotics and costly off-hour visits increase. Taken together, we conclude that the benefit of such generous subsidy is limited at least in the short-run.

Keywords: Children, Patient Cost-Sharing, Health Care Utilization, Price Elasticity, Moral Hazard **JEL codes**: 118, 113, 111

^{*} The authors thank David Chan, Liran Einav, Martin Gaynor, Jon Kolstad, Chris Muris, Krishna Pendakur, and seminar participants at the Asian and Australasian Society of Labour Economics (AASLE) Inaugural Conference 2017, Keio University, Kyushu University, National University of Singapore, Osaka University, Simon Fraser University, Singapore Management University, Stanford Asian Health Policy Program Seminar, the University of Tokyo, for their suggestions. Shigeoka gratefully acknowledges the financial support from Abe fellowship. Masaki Takahashi provided excellent research assistance. All remaining errors are our own.

[†] Faculty of Economics, The University of Tokyo, 7-3-1 Hongo, Bunkyo-ku, Tokyo 113-8654, JAPAN. Email: <u>iizuka@e.u-tokyo.ac.jp</u>

[‡] Corresponding author: Department of Economics, Simon Fraser University, 8888 University Drive, Burnaby, BC V5A1S6, CANADA, and NBER. Email: <u>hitoshi_shigeoka@sfu.ca</u>

1. Introduction

Childhood health has shown to influence both short and long-term socioeconomic outcomes and health, (Case *et al.* 2005; Currie 2009; Smith 2009), providing the ground for generous subsidy for child health care in public health insurance programs across the countries. For example, the federal government in the US regulates the share of cost paid by patients in the Children's Health Insurance Program (CHIP). Similarly, child health care is heavily subsidized in many countries with universal health insurance, including Germany, Netherland, Taiwan and Korea.¹ The lower out-of-pocket cost, however, may induces unnecessary consumption of medical services. In fact, in countries with the universal coverage, one of the few demand-side approaches on containing rising medical expenditure is to finetune the level of patient cost-sharing.²

Understanding how patient responds to price is a key to the optimal design of health insurance but past studies on patient cost-sharing are predominantly concentrated on adults and the elderly, and surprisingly little is known about children.³ We have a couple of good reasons to believe that evidence from the adults and elderly may not be simply applicable to children. First, mothers, who may need to take their children to clines, may face higher opportunity cost than the elderly. Second, nature of diseases tends to be more acute than those for the elderly (e.g., asthma vs. diabetes). At the same time, child health care utilization is often preventive and self-limiting, and hence potentially more discretionary (Leibowitz *et al.* 1987). Finally, mothers may view that the return from child health care is large as children tend to live longer.⁴

This paper examines the effect of patient cost-sharing on health care utilization among children. To do so, we newly hand-collected data on drastic expansion of subsidy for child health care in the last decade in Japan. We merge this information with individual-level monthly panel data on item-by-item health care utilization. Because municipalities expanded subsidies in different years and for different age groups, we have more than 5,000 changes in subsidy status at municipality-age-time cell, which is the level of the variation for identification in our empirical analysis. This very unique variation in subsidy generosity combined with individual panel data enables us to estimate the age-specific price elasticity in a difference-in-difference framework. Furthermore, we examine the utilization patterns by type of visit

¹ Children and adolescents are exempt from cost sharing up to age 18 in Germany and Netherlands. Similarly, children below age 3 are exempt from payment of health care in Taiwan, and those below age 6 are subsidized in Korea.

² Hossein and Gerard (2013) document that the cost-sharing for outpatient care has been increased between 2000 and 2010 among all the high-income countries examined in the study (UK, Germany, Japan, France, and the United States). ³ See Baicker and Goldman (2011) for an excellent review of the literature on patient cost-sharing. See also Chandra *et al.* (2010, 2014) for the studies on patient cost-sharing for the elderly in the US, and Shigeoka (2014) and Fukushima *et al.* (2016) for studies on the elderly in Japan.

⁴ Thus, mothers may seek health care regardless of prices or they are at least less willing to reduce their children's health care than to reduce their own (Leibowitz *et al.* 1987).

and health status to investigate whether the increases in utilization mainly reflect beneficial or low-value care.

The unique institutional background in Japan offers a clean setting in identifying patient price responsiveness since the roles of insurers and medical providers are relatively limited. As for insurances, there is no adverse selection into insurance because of universal coverage. In addition, there are no restrictions by insurers on patients' choices of medical providers and thus patients have direct access to specialist care without going through a gatekeeper or a referral system. As for medical providers, they cannot price discriminate patients as physician and hospital are paid solely based on the same national fee schedule regardless of providers' and patients' types. Relatedly, our insurance claims data *inherently* include actual transaction price which allows us to easily quantity the monetary values of (excess) utilization unlike the case in the US which is notorious for the complex price schedule.

Our paper contributes to the literature on demand for health care among children in several ways. First and most importantly, the extensive variation in subsidy status, which is always tied to the age of children, enable us to estimate age-specific price elasticity of 7–14 years-old children even at each age in month level. The price elasticity at various ages can be informative for policy makers to design more precise cost-sharing schedule.

Second, we can examine whether children *asymmetrically* respond to the price of health care since prices change in the opposite directions even *at the same age* in our setting. From a policy and welfare standpoint, understanding whether such asymmetry exists or not is crucial. On one hand, such asymmetry—if it exists—provides a cautionary note on applying the price sensitivity estimated from just one direction of price change when the policy maker considers implementing a policy with the opposite direction of price change. On the other hand, if such asymmetry does not exit, it is very useful for welfare analysis as the standard welfare calculation does not differentiate the direction of the price changes in their analysis. Despite the importance, the research design of past studies (such as randomized control trial or regression discontinuity design) precludes them from testing it as there is only single direction of price change.

Finally, our unique panel dataset which observe *both* outpatient and inpatient spending for the same individual in the same data over time allows us to examine the possibility of the cross-price effect (or known as "offset" effect in health economics)—whether beneficial *outpatient* care prevents avoidable *inpatient* admissions. More generally, if patients cut the spending for preventive outpatient care in response to price increase in outpatient care and, consequently, need to be hospitalized later, then cost-saving through reduction in outpatient care can be eventually "offset" by the subsequent increase in costly inpatient admission. To our knowledge, there is no study which examines the offset effects for children except for RAND Health Insurance Experiment (RAND HIE, hereafter) conducted in 1970s

3

(Newhouse 1993; Manning *et al.* 1987)—which randomly assign families to different patient costsharing level—but it lacks the statistical power to make any decisive conclusion (only 1,136 children).

The findings of this paper are divided into two parts. In the first part, we document the basic findings on children's price responsiveness to health care. We find that reduced cost-sharing significantly increases utilization of outpatient care among children. When the municipal subsidy lowers the patient cost-sharing from nationally-set 30% to 0% (i.e., free care), the probability of at least one outpatient visit per month increases by 6–8 percentage points (or 19–25% increases) from the mean of 32 percent in the absence of the subsidy. Similarly, total outpatient spending per month increases by 1.00–1.38 thousand JPY (roughly USD10.0–13.8), which corresponds to 22–31% increase from the mean of 4.49 thousand JPY (roughly USD44.9). The overall arc-elasticity is relatively constant for both outcomes around –0.10 throughout ages 7–14, which is somewhat smaller than conventional estimate of –0.20 for non-elderly in RAND HIE (Keeler and Rolph 1988). Interestingly, we find little evidence of asymmetric responses, meaning that children respond to different directions of price changes in a similar magnitude in our setting. This finding implies that policy makers can reasonably employ existing elasticity estimates, including ours, regardless of the direction of the price change that they consider.

The back-of-the-envelope calculation suggests that—if the full subsidy is expanded to all the municipalities among children aged 7–14 in Japan—the annual outpatient spending increases by 117 billion JPY (974 million USD). Importantly, this creates a substantial negative fiscal externality to many stakeholders: while the municipality is only responsible to cover 30% of total cost (i.e., the amount of the subsidy), the remaining 70% of the subsidy-induced excess spending must be financed by taxes and premiums, among others.

In the second part, we investigate whether subsidy-induced excess outpatient spending (moral hazard) largely reflects the increases in beneficial or low-value care. In fact, the recent work by Baicker *et al.* (2015) suggest that welfare implications of moral hazard depend on how such reduction occurs. While this is always a challenging task—especially for children as the nature of diseases tends to be acute, we examine heterogeneity of utilization patterns from various dimensions and the short-run mortality rate to answer this question to the extent possible.

We start with investigating whether we can find any evidence of increases in "beneficial" care. We find that while subsidy increases the utilization of outpatient care for the Ambulatory Care Sensitive Conditions (ACSCs)—diagnoses for which proper and early outpatient treatment should reduce subsequent avoidable admissions—there is little evidence that such increases in outpatient care translate to reduction in hospitalization by these "avoidable" conditions. More generally, we do not find any evidence of offset effects: substantial increases in outpatient spending do not accompany with the reduction in inpatient spending. Also, we find little impact on short-run child mortality while we need to

4

interpret this result with considerable caution due to very low mortality rate among children of this age range in Japan. Taken together, we find little evidence of increases in "beneficial" care.

Then, we next turn to examine whether we can find any evidence of low-value or costly care. First, we find that reduced cost-sharing substantially increases off-hour visits, validating the concern that children (and hence mothers) exploit the opportunity of generous subsidy by increasing physician visits outside of regular hours. In addition, the arc-elasticity for off-hour visits is much larger in magnitude than that of regular-hour visits for older aged children, indicating that off-hours visits seem to be more discretionary and less urgent than regular-hour visits. This increase in off-hour visits may place a substantial burden on the workload of the physicians as well as increase the overall spending since the national fee schedule sets higher fees for off-hour visits than regular-hour visits.

Second, we document that reduced cost-sharing increases the inappropriate use of antibiotics on diagnoses for which antibiotics are not recommended. This is potentially problematic as such inappropriate use of antibiotics leads to both antibiotic resistance and adverse events (Fleming-Dutra *et al.* 2016). In fact, antibiotic-resistant infections annually affect at least 2 million people and 23,000 people dies as a direct result of these infections in the United States (Centers for Disease Control and Prevention 2013). To our knowledge, no prior studies has investigated whether financial incentives, such as subsidy for child health care, increases inappropriate use of antibiotics for children.⁵

Third, we find that healthier children are more price responsive to health care than sicker children. This result suggests that health care utilization by healthy children is more discretionary and relatively low-value. The result also indicates that it is not the sickly children but the healthy children who will cutback medical care most in the absence of generous subsidy. Taken individually, each piece of evidence is not sufficient to establish the existence of wasteful utilization. But taken together, the weight of the evidence supports the notion that the generous subsidy for child health care leads to the increases in unnecessary and costly visits, implying that short-term benefit of such generous subsidy is at least far from clear.

This paper is most related to RAND HIE as our knowledge on price sensitivity to health care among the non-elderly still relies. Leibowitz *et al.* (1985) specifically analyzes children under age 13 and finds that, among others, the use of outpatient services decreased as cost-sharing increases. However, the study suffers from small sample size to identify the effect for some types of services (e.g., inpatient care).⁶ Furthermore, it is nearly 40 years old, and thus changes in practice of medicine (e.g., reliance on

⁵ For example, Foxman *et al.* (1987) examine the impact of patient cost-sharing on inappropriate antibiotic use in RAND HIE but did not separately examine for children.

⁶ The study may also suffer from the assignment of health plans that affect the entire family, so that interaction with family members may confound children's own price sensitivity. For example, if the parents receive more medical

managed care, and development of new technologies) imply that these results may not be directly applicable to the situation today, especially to countries other than the United States. A few notable exceptions from the non-experimental works are recent papers by Han *et al.* (2016) which examine the effect of cost-sharing at age 3 in Taiwan, and Nilsson and Paul (2015) which examine similar question for children in one region in Sweden.⁷ Our study may arguably entail broader policy implications than all of these studies because of variety and number of variations in cost-sharing, wider coverage of ages, and detailed analysis on comprehensive health care utilization.

Finally, this paper also is related to large literature on health insurance and child healthcare utilization especially the studies on Medicaid in the US (e.g., Currie and Gruber 1996; Gruber and Dafny 2005; Finkelstein *et al.* 2012; Goodman-Bacon 2018). However, these papers examine the effect of health insurance provision *per se* (extensive margin) rather than the effect of changes in health insurance generosity (intensive margin) like ours or RAND HIE. This distinction is important because the provision of health insurance entails large wealth effects, and thus these studies capture combined effects of price reduction and wealth. Finally, the assignment of health plans often affects the entire family like RAND HIE.

The rest of the paper is organized as follows. Section 2 provides the institutional background in subsidy to child health care. Section 3 describes the data, and Section 4 presents our identification strategy. Section 5 documents the basic findings on children's price responsiveness to health care, and Section 6 investigates whether the changes in utilization reflect beneficial or low-value care. Section 7 concludes.

2. Background

2.1. Health care system in Japan

We briefly provide the background of the Japanese health care system related to this study. Japan has a universal health insurance system since 1961, which is heavily regulated by the government. All citizens are obligated to enroll either in an employment-based insurance system or a residential-based insurance system (See, for example, Ikegami and Campbell 1995; Kondo and Shigeoka 2013). Regardless of the insurer, people face the same fee schedule and benefits package both of which are set

services thanks to insurance coverage, and hence less likely to become sick, the children who resides with parents may less likely to get cold.

⁷ Han *et al.* (2016) exploit the copayment change at age 3 in Taiwan and find that lower price significantly increases the utilization of outpatient care, especially low-value care at high-cost hospitals. Despite the increase in utilization, they find little impact on children's health. Nilsson and Paul (2015) exploit the abolishment of copayments for outpatient care among children between 7 and 19 years in one region in Sweden. They find that children increased their number of visits to a doctor while the effect vary substantially by income, with children from low income families being three times as responsive as their more advantaged peers.

by national government.

The unique institutional background in Japan offers several advantages in identifying patient price responsiveness since the roles of insurers and medical providers are relatively restricted. First, enrollment in health insurance is mandatory, and more crucially enrollees cannot choose insurers. Thus, we do not face adverse selection problem which often introduces complication in other studies. The enrollees receive identical benefits—regardless of insurance types—which include outpatient services, inpatient services, dental care, and prescription drugs. Here, note that inpatient care refers to hospital admissions with at least one overnight stay.

Second, patients face no restrictions on choices of medical providers. For example, patients have direct access to specialist care including teaching hospitals without going through a gatekeeper or a referral system unlike in the United States, where insurance companies often restrict the choices of medical providers through managed care. Third, patients cannot be price discriminated by physicians and hospitals since all fees paid to medical providers are solely based on the national fee schedule (i.e., Fee-For-Service). As a consequence, medical providers receive the same fee for the same service regardless of patient insurance type or age. This prevents so-called "cost shifting" by the medical providers in the US (Cutler 1998), where they charge private insurers higher prices to offset losses from the beneficiaries of government-funded health insurance (e.g., Medicaid). Another important implication is that any changes in health care utilization comes from quantities instead of prices since there is no room for price shopping to search cheaper providers.

2.2. Patient cost-sharing

Patient cost-sharing—for which the beneficiary is responsible out of the pocket—has been set nationally at 30% except for the following two populations: young children and the elderly. In particular, the cost-sharing is set at 20% for children below age 6. The insurer pays the remaining fraction of expenses until the beneficiary meets the stop-loss, and then patient pays a 1% coinsurance above a threshold.⁸ Unlike a typical health insurance plan in the United States, there is no deductible in Japan.

The nonlinearity imposed by the stop-loss, which is a classic but important, challenge in estimating price elasticities (Keeler *et al.* 1977; Ellis 1986). The issue is that a forward-looking patient who anticipate spending beyond the stop-loss may respond to true "shadow" price rather than "spot" price (Aron-Dine *et al.* 2015). However, this is unlikely in this setting for the following two reasons. First, only 0.067% person-month exceeds the stop-loss as the hospitalization—which is costly and main reason for reaching stop-loss—is very rare among children of this age group (only 0.28% person-month).

⁸ The supplementary health insurance that covers the remaining out-of-pocket cost does not virtually exist in Japan probably because the stop-loss prevents the catastrophic income loss upon illness.

In this sense, the spot and shadow prices are very similar. Second, the stop-loss is set monthly in Japan, unlike annually in RAND HIE and most health insurances in the United States. To the extent that illnesses are unpredictable, this shorter interval may make it harder for patients to take advantage of the stop-loss. Thus, bias coming from the nonlinearity associated with the stop-loss is likely to be negligible in our case.⁹

Importantly, many municipalities additionally provide a subsidy for child health care for those who live in the municipality regardless of their insurance type. It is called Medical Subsidy for Children and Infants (MSCI), and it has drastically expanded in last decade. Children who are eligible for subsidy at the age receive an additional insurance card, and by showing the card at medical institutions they receive the discount. Crucially, we know from our claims data the municipality of their residence, thus we can identify the level of subsidy (and hence level of cost-sharing) that each child faces. As detailed below, there is a large variation in the generosity of subsidy across municipalities and age over time, which allows us to identify its effect on health care utilization.

3. Data

3.1. Explanatory variables

Since the MSCI is set by each municipality, the level of patient cost-sharing depends on 1) where the child lives (municipality); 2) when the child visits the medical providers (time); and 3) how old the child is at the time of visit (age). The variations in these three dimensions are the sources for our identification strategy as discussed in Section 4.2.

For each municipality, we collected the following four information on subsidy for outpatient care from April 2005 to March 2015: 1) age till the subsidy is offered; 2) the level of patient cost-sharing (equivalently the level of municipal subsidy); 3) whether the subsidy is refund or in-kind; and 4) whether there are any household income restrictions for subsidy eligibility. We explain each component in detail below.

First, the generosity of subsidy is largely reflected by the maximum age till which the subsidy is provided. Note that while the eligibility age is often expressed by the school grade (e.g. till the end of junior high school), we use ages throughout this paper for convenience as the school grades are almost completely equivalent to age in Japan because of the strict enforcement of school entry rule as well as very rare grade retention and advancement.¹⁰ Second, the level and the form of subsidy (and thus cost-

⁹ In fact, even under an annual stop-loss in the US, people tend to respond myopically at spot price than future price (e.g., Keeler and Rolph 1988; Brot-Goldberg *et al.* 2017).

¹⁰ In Japan, School Education Law (SEL) obliges parents to send their children to primary schools as soon as their children turn six years of age before the school starting month, which is April. The school entry rule is strictly enforced

sharing) differ by municipality: the majority of municipalities fully subsidize child health care (i.e., coinsurance rate is reduced from the national level of 30% to 0%). Some municipalities reduce coinsurance rate to 10%, 15% or 20%, while other municipalities take the form of copayment, such as 200, 300, or 500 JPY (roughly 2, 3 or 5 USD) per visit.

Third, the payment of subsidy to patients can take the forms of either refund vs. in-kind (i.e., future vs. immediate reimbursement). While the amount of cost-sharing can be identical in two cases as long as the parents submit the required document for refund, they may prefer in-kind to refund because of time cost and/or credit constraint.¹¹ Finally, some municipalities impose income restrictions on eligibility for the subsidy. While we cannot identify the individuals who are ineligible for subsidy due to lack of income variable in our claims data, the fraction of municipalities with income restriction is very small in our data. In the empirical model, we include a dummy for income restriction.

One contribution of this paper is that we newly construct detailed information on subsidy at each municipality-age-time level (where both age and time are measured in *months*). Since information disaggregated at such a monthly level is not formally collected even by either prefectural or central government, we hand-collect through a variety of sources including municipality web page, local newspaper, and municipal ordinance.¹² Importantly, after collecting data, we directly contact each municipality and verify the accuracy of our information. Since such information for this long period (10 years) is not well kept in record in small municipalities, we limit data collection to municipalities in the six largest prefectures in Japan which result in 323 municipalities.¹³ According to national statistics, these six prefectures cover as much as 44.9% of children ages 0–15. While our data is not nationally representative, one benefit of restricting to these large prefectures is that municipalities are probably more comparable, which is useful for our difference-in-difference identification strategy.

Figure 1 plots the share of municipalities in our insurance claims data by the maximum age for the subsidy eligibility during April 2005–March 2015 among 165 municipalities—which are used in this

in Japan (only 0.03% of children are exempted from the mandatory entry). Also, the Japanese educational system is known for its social promotion system, in which automatic promotion occurs from one grade to the next without grade retention and grade advancement. Consequently, a school cohort is almost identical to a birth cohort—that is, those born in April to those born in March next year. See Shigeoka (2015) for more details.

¹¹ For example, suppose the subsidy reduces coinsurance rate from 30% to 10% at the municipality. In case of in-kind, patients only pay 10% at the medical institutions and no further action is necessary. In case of refund, patient still pay full 30% at the medical institutions, but then is reimbursed the difference between the payment and coinsurance rate, which is 20%, after filing the required documents to municipal office.

¹² While some prefectures collect such subsidy information once a year from all municipalities in the prefecture (e.g., every April), we need information on *monthly* basis for our identification strategy.

¹³ This includes municipalities that merged during this sample period. All results throughout the paper are essentially the same if we exclude them since these municipalities tend to be very small (results available upon request). There are total of 47 prefectures, and 1,719 municipalities in Japan as of January 2015.

study as explained in Section 3.3.¹⁴ Note that this figure reflects the compositional changes of municipalities as the number of municipality increases at the later period in our claims data. Importantly, within the municipalities, the subsidy expansion is always monotonic—that is, there are *no* single municipality that *lowers* the maximum age during this period (April 2005–March 2015).

The graph clearly shows that the subsidy expanded rapidly to upper ages in the last decade. For example, none of the municipalities provide subsidy till age 15 (till the end of junior high school) in April 2005, the beginning of the sample period. However, this number reaches nearly 80% in ten years by March 2015, the end of our sample period. The spike in April 2008 is explained by the fact that the central government has expanded the eligibility age for national-level subsidy (i.e., 20% coinsurance rate) from age 3 to 6 (till the start of primary school). While Figure 1 clearly shows that all municipalities in our sample have already provided the subsidy till age 6 by April 2008, this national-level subsidy expansion eases the budgetary burden on municipalities as the part of the cost is now covered by the central government. For this reason, we see the highest number of municipality-level subsidy expansions in April 2008 to ages higher than age 6 (See Appendix Figure A-1 on the precise timing of all policy changes).¹⁵

While the main reason for MSCI is to ensure the access to essential medical care for children and lessen the financial burden on parents, the exact reasons for such rapid expansion in the last decade are not fully understood. A few other justifications mentioned in the literature are to attract young couples with children for tax revenues, boost low fertility rate, and combat recent increases in child poverty (Bessho 2012). We discuss potential endogeneity of subsidy expansions in Section 4.2.

3.2. Outcome variables

Our outcome data come from the Japan Medical Data Center (JMDC), which collects and analyzes administrative insurance claims data on behalf of insurers of large corporation. Since parents of the children in the JMDC data work for large firms, our sample does not include children with extremely low household income like those who receive public assistance. Thus, the liquidity constraint is unlikely to explain the results described below. As of November 2015, the JMDC claims database contains more

¹⁴ We also collected information on subsidy for *impatient* care. However, most municipalities had already covered inpatient care till age 15 (the end of junior high school), and thus there is not much variation in eligibility of subsidy for impatient care. In fact, while we examine the effect of subsidy for inpatient care on impatient spending, we do not detect any meaningful results (results available upon request). These results are consistent with RAND HIE which finds that children only respond to price of outpatient care but not inpatient care (Newhouse 1993). Therefore, we focus on subsidy for outpatient care throughout this paper to save space.

¹⁵ While the policy changes are rather concentrated in April which is the start of the fiscal/school year in Japan, there are reasonable variations in the timing of implementation across months. This figure also demonstrates that we need a *monthly* level data on both explanatory variable (subsidy level) and outcome variable (health care utilization).

than 3 million enrollees.

JMDC data consists of administrative enrollment data and claims data. For each person, the enrollment data consist of patient ID, gender, age and the municipality of residence. The age and municipality of residence in each month are crucial in this study as the level of cost-sharing are uniquely determined by municipality-age-time. The claims data report the monthly spending, including the months of no utilization.¹⁶ Specifically, the claims data contain the year and month of the visit, and detailed information on the line-by-line medical services including diagnoses (ICD10), types of services, quantity of each service, and fees charged for each service. For example, when a prescription drug is dispensed, we have detailed information on year and month of prescription, the name of drug, ATC code, retail price, and quantity. The unit of claims data is monthly as the reimbursement to medical institutions occurs monthly in Japan. The enrollment and claims data are linked by unique patient ID.

There are a few advantages of this claims data. The biggest advantage is that the data observe both outpatient (including prescription drug) and inpatient care, *and* follow the same individual over time, which allows us to test the possibility of the "offset" effect—whether beneficial *outpatient* care prevents avoidable *inpatient* admissions in the future. In contrast, the outpatient and inpatient data are often separated in the other settings. For example, hospital discharge data do not include information on office visits and prescription drugs. Relatedly, the claims data in Japan inherently includes actual transaction prices, since the national fee schedule sets uniform prices for each procedure which is applied to all patients. This price information enables us to easily quantify the monetary values of (excess) utilization.

Our dataset is constructed in the following way. We provide the subsidy information of each municipality we hand-collected to JMDC, and then JMDC merges it with their insurance claims data inhouse by municipality and year-month, and return it to us where municipality ID and patient ID are deidentified for confidential reason. Thus, we cannot examine the heterogeneity by the characteristics of the municipality (e.g., *average* household income or maternal education) as municipality ID is scrambled. Another drawback—albeit usual for insurance claims data—is that the data do not include individual characteristics (except for gender and age of children) such as maternal education, household income, and family structure (e.g., number of kids).

As for the outcome variables, we mainly focus on two outpatient outcomes to save space: a dummy which takes one if there is at least one outpatient visit per month (an outpatient visit dummy, hereafter), and total monthly outpatient spending (outpatient spending, hereafter) measured in thousand JPY (roughly USD10). In fact, total outpatient spending in each month can be decomposed as:

Outpatient spending = $1[Visit > 0] \times (Frequency of visits | Visit > 0) \times Spending per visit - [1]$

¹⁶ The data do not, however, contain dental claims, and inpatient food and housing costs. The latter is small since the length of stay is short unlike the case of the elderly.

Thus, an outpatient visit dummy (1[Visit > 0]) captures the extensive margin of health care utilization while the outpatient spending captures the total utilization. In the online appendix, we include the results from other outcomes.

3.3. Sample restriction

We impose the sample restriction in the following ways. Our data cover the period of 10 years between April 2005 and March 2015 (120 months). We focus on ages 7–14 years-old (96 months) since as shown in Appendix Figure A-2 we do not have many observations without subsidy below age 7 and with subsidy above age 15. This happens because majority of the municipalities (81.3%) have already provided subsidy till age 6 (the start of primary school) at the beginning of our sample period, and also most of the municipalities do not provide subsidy beyond age 15 (the end of junior high school) at the end of our sample period. Therefore, we limit our sample to 6–15 years-old (one year wider on both sides of ages of interest) to identify the effect of patient cost-sharing at ages 7–14.¹⁷

Finally, we limit the sample to 165 municipalities which only have either 0% (full subsidy) or 30% (no subsidy) of coinsurance rate during our sample period for following four reasons. First, the transitions of "30% to 0%" and "0% to 30%" in cost-sharing are by far the top two variations in our data. Appendix Table A-1 lists all the combinations of transitions (including the changes from coinsurance to copayment or vice versa). These two variations accounts for 54.2% of all the transitions at municipality-age-time level, and as much as 70.0% at the person-month level. In fact, even after imposing such restriction, we still maintain as many as 5,438 changes in subsidy status at municipality-age-time cell, which is the level of the variation for identification in our empirical analysis.¹⁸ Second and relatedly, unlike other transitions, these two variations are observed at entire age ranges. This point is extremely crucial for our purpose of estimating age-specific price elasticities across wide age ranges.

Third, as shown in Appendix Table A-1, the other cost-sharing often takes the form of copayment (e.g., 500 JPY/visit), and it is inherently difficult to make them comparable to coinsurance rate (e.g., 30%). One can do this by, for example, dividing copayments by total medical expenditures. However, estimating the counterfactual number of monthly visits unavoidably introduces measurement errors.¹⁹

¹⁷ While we control for the subsidy status at ages 6 and 15 in the regressions, we do not report these estimates to save space as they are very noisy.

¹⁸ These two types of transitions are followed by "500 JPY/visit to 30%" (9.1%), "30% to 200 JPY/visit" (7.0%), and "30% to 500 JPY/visit" (5.7%), where the number in parentheses are the share at municipality-age-time level, but these transitions do not always spread across the ages.

¹⁹ We need to compute the counterfactual monthly visits for age a (in months), time t (in months), and living in municipality m. To do so, while we can keep two of them fixed, we need to relax the remaining one. For example, we may want to use the frequency of visits at age a and time t, but from other municipality m' or the frequency of visits at age a, municipality m, but from different time t'.

Because the effect of non-linear pricing on utilization is an important empirical question, we intend to investigate this further in our future research.²⁰ Fourth, it is easy to compare the asymmetric price sensitivity to the opposite directions of price changes, as detailed later. Importantly, in Appendix Table A-2, we show that the characteristics of children as well as their health care utilization are quite similar between 165 municipalities in our data and the remaining municipalities. This alleviates the concern that our sample municipalities are specific, and thus the results are not generalizable.

3.4. Descriptive statistics

Table 1 provides the summary statistics of selected variables in our sample at municipality, individual, and person-month levels in Panels A, B, and C respectively. Panel A shows that each municipality is observed on average 76.6 months, and 68.5% of the municipalities have at least one subsidy expansion. Importantly as discussed in Section 4.1, the source of variation for identification does not simply come from the *expansion* of subsidy but also *expiration* of the subsidy at certain age. At the individual level (Panel B), we have total of 63,590 individuals, and each individual is observed on average 36.2 months, and 21.8% of individuals experience at least one subsidy change. Concretely, 16.5% experience at least one subsidy expansion (from 30% to 0%) and 19.3% experience at least one expiration (from 0% to 30%) as detailed in the Section 4.1. Gender is well balanced (48.8% is female). Finally, Panel C reports some key variables at the unit of our analysis (person-month). We have a total of 2,303,335 person-month over the sample period of 120 months. Almost all the subsidy is provided in the form of in-kind (99.9%), and very few municipalities impose income restriction for eligibility criteria (1.5%). In terms of utilization, 40.7% of children make at least one outpatient visit per month on average, and spend 6.09 thousand JPY (roughly USD60.9) per month including zero-spending, and 14.95 thousand JPY (roughly USD149.5) conditional on at least one visit. Inpatient admission for this age range is very low (only 0.28%) but inpatient care is much more costly when admitted (406.52 thousand JPY or USD4065.2) than outpatient care.

The simple plots of raw data already reveal interesting patterns. Panel A of Figure 2 plot the raw means of outpatient utilization at each age for children who live in municipalities with subsidy (labeled "subsidized") and those who live in the municipalities without subsidy (labeled "no subsidy"). The graph on the left for an outpatient visit dummy shows that the line with subsidy is always higher than the line without subsidy by 8–11 percentage points at any age ranges, while both age profiles is declining since the average health may improve at older ages. The graph on the right also demonstrates a similar pattern for outpatient spending: the mean outpatient spending is roughly 2 to 3 thousand JPY (USD20–

 $^{^{20}}$ A number of studies point out the possibility that there may be special psychological properties to zero price (e.g., Shampanie *et al.* 2007).

30) higher with the subsidy than without subsidy or by 40–60% higher, which is substantial. While this figure does not account for compositional changes in the sample, the main message from the regression analysis below is similar. Panel B of Figure 2 plots the age profile of impatient outcomes, which are aggregated in age in years as hospital admission is a very rare event in this age group. In contrast to outpatient outcomes, we see no clear difference in inpatient admission dummy and inpatient spending with and without subsidy.

Appendix Table A-3 lists the major diagnosis groups in our sample by ICD10. The largest share comes from diseases of the respiratory system, which account for roughly one third of the all diagnoses. We also list the top 10 individual diagnoses at the ICD10 4-digit level. Importantly, the top ranked diagnoses tend to be *acute* such as acute bronchitis, and acute upper respiratory infection compared to the elderly who tend to have more *chronic* diseases.

4. Identification Strategy

4.1. Source of variations in patient cost-sharing

Before presenting our estimation equation, it is important to clarify the two sources of variations used in our identification strategy. Importantly, the subsidy (hence patient cost-sharing) is uniquely determined by municipality, age, and time. Put differently, each cohort (defined by municipality and birth in months) experience own price schedule as long as they do not move across municipalities. Figure 3 illustrates one example of patient cost-sharing schedule in a particular municipality. By drawing the two separate price schedules for two cohorts that are just born one month apart, we demonstrate our source of variations in subsidy status at different ages as well as the concept of asymmetry in price changes.

Panel A draws the price schedules for each cohort *before* subsidy expansion. The solid line draws the price schedule for cohort born in July 1998 ("younger" cohort, hereafter), and the dotted line for cohort born in June 1998 ("older" cohort, hereafter), born a month before the younger cohort. Suppose that the municipality provides full subsidy (i.e., 0% coinsurance rate) till the beginning of primary school (roughly age 6). Since the school year starts in April in Japan, the younger cohort is 6 year and 9 months old, while the older cohort is 6 years and 10 months old, when both cohorts enter primary school in April 2005. Above this age, children pay national level of 30% coinsurance rate.

Suppose that in October 2007 the municipality expands the subsidy up to the end of junior high school (roughly age 15). Panel B draws the price schedules *after* subsidy expansion. The younger cohort (solid line) pays full 30% from age 6 years and 9 months to age 9 years and 2 months, a month before subsidy expansion in October 2007. Thanks to subsidy expansion, the cohort enjoys the free care from

age 9 years and 3 months till age 15 years and 8 months when the cohort graduates from junior high school in March 2014. Then, once again the cohort pays full 30% after age 15 years and 9 months. On the other hand, the price schedule for older cohort (dotted line) is shifted by one month to the right as the cohort is one month older than younger cohort at the entry of primary school, the subsidy expansion, and graduation from junior high school.

There are two important points to make from this simple illustration. First, any cohorts between 6 and 15 years old are benefited from the same subsidy expansion. As a result, each cohort uniquely experiences the subsidy expansion as well as expiration at different ages. This enables us to estimate the price elasticity for broad age ranges (ages 7–14) even at the monthly level. Price elasticities at different ages can be informative for the government to design more flexible cost-sharing schedule.

Second, we can investigate *asymmetric* responses to the direction of the price changes as our variation includes the price changes in both directions even *at the same age*. It is not a priori obvious whether elasticities are the same by the direction of price changes. On one hand, the elasticities can be smaller in price increase ("worse") than price decrease ("better") if the exposure to free care before the price increase leads to habit formation of visiting doctors, and hence the utilization does not decrease much even after the price increase. On the other hand, the elasticities can be larger in price increase ("worse") than price decrease more to the loss of subsidy ("worse") than the gain of subsidy ("better") due to loss aversion. Hence, this is ultimately an empirical question.

The research design of past studies (such as randomized control trial in RAND HIE or regression discontinuity design) do not allow them to test this question because there is only single direction of price change. Fortunately, in our setting, out of 5,438 changes in subsidy status at municipality-age-time cell, the directions of price changes are split nearly half: 2,505 changes (46.1%) are the expansion of the subsidy (labeled "better" or 30% to 0%) while 2,933 changes (53.9%) are the expiration of the subsidy (labeled "worse" or from 0% to 30%).²¹

4.2. Identification strategy

We exploit the unique variation in subsidy for child health care across municipality, age, and time combined with the longitudinal claims data in difference-in-difference framework. Specifically, our basic estimation equation is:

 $Y_{iamt} = \alpha + \sum_{a=85}^{179} \beta_a subsidized_{iamt} + \gamma X'_{mt} + \delta_a + \varphi_m + \pi_t + \theta_i + \varepsilon_{iamt} - [2]$ where Y_{iamt} is the health care utilization by a child *i* whose age is *a* (measured in months), in time *t* (year-month), and living in municipality *m*. *subsidized*_{iamt} is a dummy, which takes one if the

²¹ Note that since we only allow for the municipalities to have either 0% or 30% of coinsurance rate *throughout* our sample period, the actual number of these two transitions is slightly smaller than those listed in Appendix Table A-1.

outpatient care for children is fully subsidized at age *a*. Since children become eligible or ineligible for the subsidy at the beginning of the specified month, we can assign the subsidy dummies using the age in months without measurement errors. δ_a , φ_m , π_t are fixed effects for age, municipality, and time respectively. The simple illustration in the previous subsection heightens the importance of controlling for these fixed effects. Also, η_i is the individual FE which captures the unobserved time-invariant characteristics of patients and addresses the compositional changes of individuals in the unbalanced panel data. We also control for two time-varying municipality variables: a dummy that takes one if subsidy is in-kind rather than refund and a dummy takes one if there exists income restriction on eligibility for the subsidy while recognizing the lack of the statistical power to identify these effects (thus not the focus of the paper). Standard errors are clustered at the municipality to account for serial correlation in the error terms within the municipalities.

While we can technically estimate β_a (age *a* in months), as shown later, the monthly estimates β_a are relatively stable within age in years. Therefore, we instead report β_A (age *A* in years) as below to obtain more statistical power without losing much information:

$$Y_{iamt} = \alpha + \sum_{A=7}^{14} \beta_A \{subsidized_{iamt} \times 1(Age A)\} + \gamma X'_{mt} + \delta_a + \varphi_m + \pi_t + \theta_i + \varepsilon_{iamt} - [3]$$

where 1(Age A) is an indicator variable which takes one if the person is more than age *A* but less than age A + 1 (or equivalently $1(Age A) = 1(A \le a < A + 1)$). We construct age in year dummies in this way so that age corresponds to school grade.²² Our coefficients of interest are series of β_A (A = 7-14) which captures the average effect of subsidy within the age ranges.²³ Importantly, we still include δ_a at the monthly level to account for any age in month specific effects.

For our main analysis, we focus on the sample of non-movers (98.3% of the sample) who stay in the same municipality while they are observed in our data. The migration rate in our sample is lower than the actual migration since *intra*-municipality migration is not counted since the subsidy level is still the same.²⁴ Although we have very few movers in our data, we are still concerned that the estimated effects of subsidy may be biased if sicker children move to a municipality that offers a generous subsidy. To alleviate this concern, in Appendix Section L, we estimate a discrete choice model that examines whether children (and their parents) migrate to a municipality that provides free child health care,

²²For example, age 6, 12, and 15 corresponds to age just before start of primary school, the last year of primary school, the last year of junior high school in Japan, respectively.

²³ We abstract from whether this effect stems from the patient-induced demand, that is, children or mothers ask more care when price is low, or physician-induced demand, that is, doctors may provide aggressive treatments stemming from their economic motives/benevolence. See, for example, Iizuka (2007, 2012) for studies that attempt to disentangle these two effects.

²⁴ Furthermore, the inter-municipality migration is much more common before the children enter primary school, which is outside of our sample age ranges.

finding little evidence that supports such a migration pattern. In addition, we report that including movers into the sample hardly change the results because of small number of inter-municipality migration. For non-movers, since φ_m and θ_i are identical, our final estimation equation is written as²⁵:

$$Y_{iamt} = \alpha + \sum_{A=7}^{14} \beta_A \{subsidized_{iamt} \times 1(Age A)\} + \gamma X'_{mt} + \delta_a + \pi_t + \theta_i + \varepsilon_{iamt} - [4]$$

In addition, we can exploit the two opposite directions of changes in subsidy status to examine the asymmetric price responses of children. Thus, a subsidized dummy in equation [4] is decomposed to two sets of dummies as:

$$Y_{iamt} = \alpha + \sum_{A=7}^{14} \beta_A^{Better} \{subsidized_{iamt} \times better_{iamt} \times 1(Age A)\} + \sum_{A=7}^{14} \beta_A^{Worse} \{subsidized_{iamt} \times worse_{iamt} \times 1(Age A)\} + \gamma X'_{mt} + \delta_a + \pi_t + \theta_i + \varepsilon_{iamt} - [5]$$

where *better*_{*iamt*} is an indicator equal to zero before subsidy is not available, and equal to one in all period after the subsidy is introduced, even if the subsidy expires. Similarly, *worse*_{*iamt*} is an indicator variable equal to zero before subsidy expires, and equal to one in all years after the subsidy expires, even if the subsidy becomes available.²⁶ Because of the way that the indicators are defined, β_A^{Better} tests the effect of *decrease* in cost-sharing on utilization, relative to individuals in other municipalities without subsidy, after the subsidy is expanded, relative to the period when the subsidy was not available; on the other hand, β_A^{Worse} tests the effect of *increases* in cost-sharing on utilization, relative to the period when the subsidy was not available; on the other municipalities with subsidy, after the subsidy is expired at the age, relative to the period when the subsidy was available.

The identifying assumption in our difference-in-difference strategy is that there are no unobserved municipality specific changes that (1) are correlated with changes in subsidy in the municipality and (2) are correlated with municipality-specific changes in the health care utilization. Since the level of variation for both age and time are measured in months and thus there are as many as 5,438 changes in subsidy statuses at municipality-age-time cell, it is difficult to imagine that such omitted variables are likely to influence our estimates. Nonetheless, it is still possible that the municipality with different pretrend in utilization may implement the subsidy expansion at the different timing, which may bias our estimates. For example, if the municipalities in better business cycles are more likely to implement the subsidy expansion, while income effects simply increase utilization, our estimates can be upward biased.

²⁵ For non-movers, since *time* = (*birth* + *age*), controlling for age and time FEs essentially determines the cohort (i.e., month of birth), which experiences the same patient cost-sharing schedule.

²⁶ Currie *et al.* (2015) employ a similar strategy to examine the asymmetric effects of opening and closing of toxic plants on housing values. Note that this way of constructing variables in our data only makes sense when the changes of subsidy status within the individual are less than or equal to two. Thus, in the analysis of asymmetric price response, we remove 921 individuals (1.45%) who experience more than two changes in subsidy status. We confirm that our baseline estimates are essentially unchanged once we remove these individuals from the sample (results available upon request).

To account for such concern, we take three approaches. First, we conduct the event-study that normalize the data to the timing of the subsidy changes, and examine whether there are any systematic differences in pre-trend between treated and control municipalities before the changes. Second, we add municipality specific time trend and even time-by-municipality FEs (where time is measured in months), to examine the robustness of our baseline estimates. The latter specification is most stringent as these fixed effects capture any municipality specific policy changes or events in a particular month, if any, such as income transfers, other subsidies or business cycles. Finally, we limit our sample to only those individuals who experienced at least one change in subsidy status. Since we only exploit the *timing* of the changes in subsidy status in this specification, we can to some extent mitigate the concern that individuals in the treatment and control municipalities might be different.

5. Basic Results

5.1. Graphical evidence: Event-study

Before presenting the regression results, we provide the graphical evidence on changes in outpatient outcomes in the form of an event-study. Here, we normalize the data around the change in subsidy status for any ages to increase statistical power. Then, we replace the subsidized dummy in estimation equation [4] by a series of dummy for each month ranging from 12 months prior to the change in subsidy status to 12 months after the change (T= –12 to +11, where T=0 is the change in subsidy status). Thus, the estimates are the weighted average of treatment effects across all ages.

Figure 4 presents the results of event-study for an outpatient visit dummy (Panel A), and outpatient spending (Panel B), separately for subsidy expansion ("better") and subsidy expiration ("worse"). The reference month is three months before the change in subsidy status (T= -3). Note that the scales of *y*-axis are set the same within the panels so that two figures for opposite directions of subsidy changes are visually comparable.

There are a few important points to make from these graphs. First, there does not seem to show any pre-trend as the estimates are mostly close to zero before the changes in subsidy status. We are very reassured as this result addresses the concern that the municipalities that expand the subsidy may have different trend in health care utilization than municipalities that do not.

Second, there is substantial anticipatory utilization as indicated by drops in subsidy expansion ("better") and surges in subsidy expiration ("worse") just before T=0. This pattern reveals that some children (and hence mothers) are well-aware of the changes and behave strategically by delaying visits till the subsidy becomes available, and conversely rush to visit just before the subsidy is over. On one hand, the existence of the anticipatory utilization is rather surprising as nature of diseases for children

tend to be acute. On the other hand, the fact that the magnitude is larger for subsidy expiration ("worse") than subsidy expansion ("better") indicates that at least a part of these visits is indeed acute because one cannot delay treatments too much till the expansion while one can more easily stockpile before expiration.²⁷ Also, the differential response can be behavioral in that mothers of children may react more to price increase rather than price decrease (loss aversion).²⁸ Importantly, as we include age and time FEs (both in months), this difference is not driven by a particular age or month effect such as the expiration of subsidy after graduation from primary school. In any case, since such anticipated effects—which may overstate our estimates—seems to be concentrated within two months from T=0, we exclude these four months of the data throughout the paper. For instance, a similar approach is taken by Chandra *et al.* (2010). In fact, as shown later, the estimates and hence elasticities are barely affected after removing more than two months from T=0.²⁹

Finally and most importantly, the effect on utilization seem to be permanent rather than transitory since the level of the utilization after T=0 does not revert to the level before T=0. This result justifies the use of the difference-in-difference strategy as we do not need to rely on observations only around T=0 to estimate the effect of cost-sharing on utilization. In fact, Appendix Figure C-1 further expands the window of the event-study to ± 24 months (instead of ± 12 months) from the subsidy changes. It is reassuring that the estimates are relatively stable even 24 months (2 year) after the change in subsidy status. Technically, we can do the event-study for each age despite the loss of the statistical power. Appendix Figures C-2 shows the event-study at age 9 and 12, as examples. The fact that we observe opposite signs of changes in utilization for subsidy expansion and expiration even *at the same age* indicates that we are unlikely to capture something that is particular at that age (e.g., finishing primary school at age 12).

²⁷ In fact, Appendix E shows that while we see anticipatory utilization for all service categories examined (medication, consultation fees, laboratory tests, and non-surgical procedure), the magnitude of anticipatory spending seems to be larger in medication than non-surgical procedure.

²⁸ Another potential explanation is information availability; that is, on average people may be more aware of subsidy expiration ("worse") than subsidy expansion ("better"). Suppose, the free care is expanded from age 6 to 15. Then, a 6-year old child have at the maximum of 9 years to be aware of the end date of subsidy while a nearly 15-year old child have at the minimum of 1 month. Assuming the uniform distribution of children across ages, they have on average 4.5 years from the start of the subsidy expansion to realize the end date. On the other hand, while the subsidy expansion should be announced at least a few months (or even longer) in advance, children have less time to know the start date of expansion.

²⁹ To the extent that the *net* change in utilization around the changes in subsidy status is positive, the excess mass of anticipatory utilization (such as delayed treatment) can be potentially viewed as a particular form of moral hazard (Cabral 2017). If so, the estimates and the corresponding elasticities without removing the data may provide the upper bound.

5.2. Main results

Figure 5 demonstrates the graphical presentation of estimating equation [4] which plots β_A for each age (A=7-14) in the upper half, and the corresponding arc-elasticity from equation [5] in the lower half. Panels A and B present the results of an outpatient visit dummy, and outpatient spending respectively. Note that Appendix Figure A-3 plots the monthly estimates (β_a) instead of yearly estimates (β_A) . Since monthly estimates are relatively stable within age in year (and statistically significant at 1%) level for any age in months), we do not lose much information by reporting β_A .

Panel A of Figure 5 reveals that the estimates (β_A) on an outpatient visit dummy are relatively stable across ages 7–14, and are statistically significant at the conventional level for any ages. With subsidy, the probability of seeing a doctor at least once a month increases by 6-8 percentage points higher than the probability without subsidy. This translates into 19-25% increases from 0.32, the mean without subsidy among ages 7–14. The corresponding arc-elasticities presented at the bottom half range from -0.07 to -0.10.³⁰ While the considerable caution is necessary to compare the elasticities estimated across countries and time periods, the arc-elasticities we find are smaller than the gold-standard estimate of -0.17 to -0.31 for non-elderly from the RAND HIE (Keeler and Rolph 1988), and similar to Han et al. (2016), which document the arc-elasticities of -0.12 and -0.08 for regular and emergency outpatient care among 3-year-old children in Taiwan. See Appendix Section B for more detail on these elasticities and related comparisons with literature.³¹

Panel B of Figure 5 plots that estimates of outpatient spending. While the outpatient spending is arguably of the greatest interest—as it eventually captures the size of total utilization—the estimates are less precise than extensive margin documented above because of skewed right tail of spending. The estimates are slightly declining as one gets older: with subsidy, the outpatient spending increases by 1.38 thousand JPY (13.8 USD) per month at age 7, and by 0.998 thousand JPY (9.98 USD) per month at age 14 than those without subsidy. These estimates correspond to 18–31% increases from 4.49 thousand JPY, which is the mean value for ages 7-14 without subsidy. Since the mean outpatient spending—which is

estimate on price elasticity can be lower bound. Unfortunately, we don't have any data on waiting time. ³¹ The arc-elasticity for each age in year A is defined by $\varepsilon_A = \left(\frac{Q_{1A}-Q_{0A}}{Q_{0A}+Q_{1A}}\right) / \left(\frac{P_{1A}-P_{0A}}{P_{1A}+P_{0A}}\right) = \left(\frac{\beta_A}{Q_{0A}+Q_{1A}}\right) / \left(\frac{0-0.3}{0.3+0}\right) = \frac{\beta_A}{Q_{0A}+Q_{1A}}$

³⁰ If the congestion at medical institutions deter some demand of health care in order to avoid the waiting cost, our

 $^{-\}frac{\beta_A}{Q_{0A}+Q_{1A}}$ where subscripts 0 and 1 denotes unsubsidized and subsidized, respectively. β_A are the estimates from the equation [4]. We report arc-elasticity because they are widely used and comparable to estimates from RAND HIE, in which the largest plan was also the free care plan. However, one can make an argument that when the starting price is zero, as in our case, price elasticity may not be well defined. Thus, as an alternative, we also report the semi arcelasticity, which is defined by $\varepsilon_A^{semi} = \left(\frac{Q_{1A}-Q_{0A}}{Q_{0A}+Q_{1A}}\right) / \left(\frac{P_{1A}-P_{0A}}{2}\right) = \left(\frac{\beta_A}{Q_{0A}+Q_{1A}}\right) / \left(\frac{0-0.3}{2}\right) = \varepsilon_A / 0.15$. Thus, in our case, *semi* arc-elasticity is simply arc-elasticity divided by 0.15. For example, semi arc-elasticities are reported in Brot-Goldberg et al. (2017), and Nilsson and Paul (2015).

the denominator of arc-elasticity $(= -\beta_A/(Q_{0A} + Q_{1A}))$ —is relatively stable across ages, arc-elasticities are largely governed by the sizes of β_A . Hence, arc-elasticities are also slightly declining in ages, which go from -0.12 at age 7 to -0.08 at age 14.

Since the total number of children aged 7–14 in Japan is roughly 8.8 million according to the Population Census in 2015 (Statistics Bureau 2015), the back-of-the-envelope calculation suggests that —if the free care is expanded to all the municipalities in Japan—the annual outpatient spending increases by 117 billion JPY (974 million USD).³² It should be noted that although municipalities bear 30% of the subsidy-induced spending, the remaining 70% of increased spending should be borne by others. Because the universal health coverage is financed by taxes (39%), premiums (49%), and out-ofpocket (12%) in Japan (Ministry of Health, Labour and Welfare 2014), the municipal subsidy has a substantial negative fiscal spillover on many stakeholders.

Table 2, which corresponds to Figure 5, also shows that the estimates on a dummy for in-kind payment, and a dummy for income restrictions take the expected sign: in-kind payment increases the outpatient visits by 4.7 percentage points, which is more than half the size of the estimates in patient cost-sharing from 30% to 0%, and decrease by 2.0 percentage points when there is income restriction on eligibility for the subsidy.³³ For outpatient spending, the signs of these coefficients are also as expected, but not statistically significant at the conventional level.

We subject these results to a series of robustness checks. In the interest of brevity, we leave the detailed descriptions of the exercises to Appendix Section D. Critically, the results in Table 2 on the causal effects of patient cost-sharing are robust across all specifications considered. In particular, we address the potential concern that our control group—namely children in municipalities without changes in subsidy—exhibits a different time trend than children in municipalities with subsidy changes. Since the estimates in event-study before T=0 seems to be reasonably smooth and close to zero, this does not seem to be a serious concern. Nonetheless, we add the time-by-municipality FEs (where time is measured in months) to account for the time-varying municipality characteristics that can be potentially correlated with both the expansion of the subsidy and health care utilization. We are reassured that the estimates are barely changed. We also collapse the data at municipality-age-time cells, which is the level of variation, to partially account for zero spending but the estimates are almost identical. As a separate exercise, we present the sensitivity of our estimates to the size of the "donut-hole". The estimates and hence elasticities are barely affected after excluding 2 months from both sides of T=0.

³² We multiply each β_A by the number of children in the age *A* in 2015 and sum them up to calculate monthly spending. Then, we multiply it by 12 to convert to annual spending. The average exchange rate of 120.39 JPY/USD in March 2015 is used.

³³ This result is consistent with Zhong (2011) which shows that immediate reimbursement increases health care utilization compared to future reimbursement in China. Despite its importance, the effect of reimbursement method on utilization is understudied in health economics.

In Appendix Section E, we also report the results on frequency of outpatient visits and the outpatient spending per visit as outcomes. Interestingly, while both spending and frequency of visits increase, the spending *per visit* is also positive and statistically significant at the conventional level. This result suggests that the increases in total outpatient spending are driven by both the increases in the frequency *per se* and spending per visit. A simple decomposition indicates that the increases in frequency of visits accounts for 70.2–81.0% of increases in total outpatient spending.³⁴

In Appendix Section F, we also examine each type of medical services: the medication accounts more than half the share of total spending (54.1%), followed by consultation fees (18.4%), laboratory tests (17.2%), and non-surgical procedure (5.3%).³⁵ The consultation fees—which are charged in each visit and thus is closely related to the frequency—are least price sensitive. On the other hand, the medical services related to the treatment intensity, specifically laboratory tests (including imaging) and non-surgical procedures, are more price sensitive. This result is consistent with our finding that not only the frequency of visits but also the spending per visit increases. Interestingly, the medication is not as price sensitive as other service categories.

5.3. Asymmetric responses

In this subsection, we investigate whether children asymmetrically respond to price of health care by exploiting the unique variation of price changes in opposite directions. Figure 6 demonstrates the graphical presentation of estimating equation [5] which plots β_A^{Better} and β_A^{Worse} for each age (A=7-14) in the upper half, and the corresponding semi *point*-elasticity in the lower half.³⁶ Panels A and B report the results of an outpatient visit dummy, and outpatient spending, respectively. Table 3 corresponds to Figure 6.

Two things are worthy of mentioning. First, the estimates take completely opposite signs for opposite directions of changes in subsidy status, reassuring that our estimates are not driven by just one direction of price change. For both outcomes, the estimates are statistically significant at 1% level at any age ranges. Second, while the semi point-elasticity for an outpatient dummy is slightly larger in magnitude for subsidy expansion ("better") than subsidy expiration ("worse") at some ages, we do not

³⁶ For the analysis of asymmetric responses, we instead report semi *point*-elasticity instead of *arc*-elasticity as we exactly know the starting quantity, and also the direction of the price changes. Specifically, the semi *point*-elasticity for each direction of price changes are defined as: $\varepsilon_A^{Better} = \left(\frac{Q_{1A}-Q_{0A}}{Q_{0A}}\right)/(P_{1A}-P_{0A}) = -\left(\frac{\beta_A^{Better}}{Q_{0A}}\right)/(0.3, \varepsilon_A^{Worse}) = \left(\frac{Q_{0A}-Q_{1A}}{Q_{1A}}\right)/(P_{0A}-P_{1A}) = \left(\frac{\beta_A^{Worse}}{Q_{1A}}\right)/(0.3, where \beta_A^{Better})$ and β_A^{Worse} are estimates from equation [5].

³⁴ We simply calculate the percentage increase from the mean for both frequency and spending per visit, and calculate the fraction of the first one divided by the sum of the two.

³⁵ Note here that medication includes fees not only for medicine itself but also related to prescribing and dispensing medications, including fees at the pharmacy.

find much difference in semi point-elasticity for outpatient spending.³⁷ Since we are eventually interested in the overall spending, we conclude that there is little evidence of asymmetric price responses.

The nonexistence of the asymmetry at least in this setting has an important implication as the price sensitivity estimated from one direction of price change may be applicable even if the policy maker wants to implement the policy with the opposite direction of price change. Also, it is very useful for welfare analysis as the standard welfare calculation does not differentiate the direction of the price changes. Since we see little asymmetry in price responsiveness for our baseline results, we focus on the estimates from equation [4] without asymmetry hereafter.

6. Beneficial or low-value care

The remaining important question is whether the increased outpatient utilization due to lower price (moral hazard) largely reflect beneficial or low-value care. In fact, the recent work by Baicker *et al.* (2015) suggest that welfare implications of quantity changes depend on how they occur. While this is always a challenging task—especially for children as the nature of diseases tends to be acute, we examine heterogeneity of utilization patterns from various dimensions (such as type of visit and health status) and short-term child mortality to answer this question to the extent possible. We fully recognize that subsidy-induced utilization should include some aspects of both essential and non-essential care, as documented in RAND HIE (Newhouse 1993; Manning *et al.* 1987). Thus, we focus on the relatively extreme cases of utilization patterns to tackle this question.

6.1. Evidence of beneficial care

We start with investigating whether subsidy-induced care clearly benefits children. For this, we examine whether increases in outpatient care prevents avoidable inpatient admissions and reduces short-term mortality.

6.1.1. Ambulatory Care Sensitive Conditions (ACSC)

We begin our analysis by examining whether outpatient care prevents avoidable inpatient admissions. Instead of looking at broad disease categories or choosing them arbitrary, one useful set of preventive care is the utilization for so-called Ambulatory Care Sensitive Conditions (ACSC) or "avoidable conditions"—diagnoses for which timely and effective outpatient care can help to reduce the

³⁷ For outpatient spending, while the magnitude of numerator in semi point-elasticity is larger for "worse" (β_A^{Worse}) than "better" (β_A^{Better}), the denominator is also larger for "worse" (Q_{1A}) than "better" (Q_{0A}) for an obvious reason. Thus, the resulting semi point-elasticity is similar in both directions.

risks of hospitalization by either preventing the onset of an illness or condition (e.g., asthma).³⁸

We employ the list of ACSC from Gadomski *et al.* (1998) that specifically focus on children.³⁹ Appendix Table G-1 provides the lists of the ACSC with corresponding ICD10 codes, and the fraction of each ACSCs in our sample. Column (2) indicates that conditional on visit, as much as 41% belongs to ACSC which verifies the acute nature of diseases for children. Among the list of 17 ACSCs, severe ENT infections (56.9%) and asthma (31.5%) stand out, and account for nearly 90% of total ACSCs.

Therefore, to save space, Figure 7 plots the estimates from equation [4] when outcome is an outpatient visit dummy for (i) any ACSC, (ii) severe ENT infections, and (iii) asthma.⁴⁰ Panel A shows that outpatient visits for these diagnoses increase when subsidized, and all the estimates are statistically significant at the conventional level. Table 4 reports the estimates that are plotted in Figure 7. For example, the outpatient visit by any ACSC increases by 2–4 percentage points during ages 7–14 where the mean without subsidy is 0.11. The corresponding arc-elasticities presented in Appendix Figure G-1 are roughly –0.1. Since nearly 40 percent of the diagnoses for this age group are ACSCs, the arc-elasticities are very similar to overall arc-elasticities shown in Figure 5.

These results at a glance seems consistent with the literature that people are not only price sensitive to non-essential or low-value care but also to essential care. For example, RAND HIE documents that price sensitivity for preventive care is similar to that for acute or chronic care among children (Leibowitz *et al.* 1985). However, most of past studies could not examine whether such seemingly beneficial care indeed lead to better health of patients or prevent avoidable hospital admissions. Here, one big advantage of our insurance claims data is that it includes the information for both outpatient and inpatient care from the same individual over time unlike most existing datasets that only capture either outpatient care or inpatient care. Thus, we can directly examine whether such increases in preventive care at outpatient setting indeed lead to the reduction in hospitalization.

Panel B in Figure 7 plots the estimates from equation [4] where the outcomes is an *inpatient* admission dummy while the explanatory variables are subsidy status for *outpatient* care as before.⁴¹ We do not see any declines in the hospitalization associated with any ACSC or other individual ACSCs. Since the hospitalization among children are very infrequent (0.28% of all person-months), the estimates are overall imprecise. However at least the point estimates are always positive instead of negative and

³⁸ Nonetheless, we estimate equation [4] by the broad diagnosis groups as indicated in Online Appendix Table A-2. We do not find much heterogeneity except for the "Injury, poisoning and certain other consequences of external causes" which shows slightly smaller arc-elasticity as these conditions may be more urgent and less discretionary (results available upon request).

³⁹ For example, Kaestner *et al.* (2001) and Dafny and Gruber (2005) examine the ACSC for children.

⁴⁰ Unfortunately, insurance claims data in Japan lists all the ICD10 that are diagnosed in the month instead of diagnosis for each visit, making it difficult to examine the other outcomes such as spending and frequency of visits by ICD10.
⁴¹ Whenever we examine the inpatient outcomes, we also control for the subsidy for inpatient care while the adding

these variables does not affect our results as there is little variation in inpatient subsidy.

even statistically significant at some ages in case of asthma. Furthermore, since there is a possibility that the benefit of preventive care may emerge with a lag, we also estimate a variant of equation [4] where the explanatory variables are lagged outpatient subsidy dummies in a simple dynamic model. We find little evidence of any lagged effects either (results available upon request).

6.1.2. Offset effects

More generally, we can examine whether *outpatient* spending replaces *inpatient* spending widely known as "offset" effect in health economics. On one hand, if the outpatient visit is preventive and beneficial in that it leads to the detection and successful treatment of a condition that would have otherwise resulted in hospitalization, it will *decrease* inpatient care use. On the other hand, if the outpatient visits lead to a referral to a specialist for additional examination and invasive treatment for a condition that would have otherwise resolved itself in time (self-limiting), it will *increase* inpatient care use. Note that the analysis on ACSCs in the previous subsection is a special case of the "offset" effect, which focuses on conditions specific to ACSCs.

Whether outpatient care is a substitute or complement for inpatient care is an important but unsettled question in health economics. Overall, RAND HIE does not find the evidence of "offset" effects (Newhouse 1993). Some studies report that outpatient and inpatient care are rather complements (e.g., Kaestner and Lo Sasso 2015) while a few studies that document the evidence of offset effects are concentrated among the elderly (e.g., Chandra *et al.* 2010, Trivedi *et al.* 2010). To our knowledge, there is no study which examines the cross-price effects for child health care except for RAND HIE which lacks in statistical power due to very small sample size of children (1,136 children whose families participated in a randomized trial).

To investigate this question, we replace the outcome in equation [4] by inpatient dummy or inpatient spending while explanatory variable is still the subsidy for outpatient care. In this way, we can investigate whether the change in subsidy for outpatient care has any impact on impatient care.⁴²

Panel A of Figure 8 plots the estimates on the probability of hospitalization, and Panel B plots the estimates on inpatient spending. Table 5 shows that out of 16 estimates (8 for each age for each outcome), none of the estimates except for one are statistically significant at the conventional level. Also,

 $\sum_{A=7}^{14} \beta_A \{ outpatient \ spending_{iamt} \times 1(Age \ A) \} + \gamma X'_{mt} + \delta_a + \pi_t + \theta_i + \varepsilon_{iamt} \ \text{where}$

⁴² Following Kaestner and Sasso (2015), we also directly estimate the effect of outpatient spending on inpatient use in the instrumental variable approach. To account for potential endogeneity of outpatient spending, we instrument outpatient spending by outpatient subsidy. Specifically, we estimate: *inpatient care_{iamt}* = α +

[{]outpatient spending_{iamt} × 1(Age A)} are instrumented by {subsidized_{iamt} × 1(Age A)}. The outcome is either inpatient spending or inpatient admission dummy. In this sense, the results presented in Figure 8 are the reduced-form estimates. We find that the estimates β_A are always positive as in Figure 8, but they are not precisely estimated (results available upon request).

the estimates are mostly positive albeit statistically insignificant, which imply that the generous subsidy for outpatient care does not seem to reduce the utilization of inpatient care, despite the large increases in the outpatient visits and spending documented so far.

We also confirm that directions of price changes do not matter: we do not find evidence that reduced use of outpatient services due to subsidy expiration leads to increased demand for impatient services nor evidence that increased use of outpatient services due to subsidy expansion leads to reduced demand for impatient services. Furthermore, we find little evidence of lagged effects (all results available upon request). These results echo with the findings on ACSC in the previous subsection. Of course, hospitalization is just one short-term indicator of worsening health, and we cannot rule out the possibility of long-term positive return on health. At least in our data, we do not find any evidence of "offset" effects on inpatient care despite the substantial increases in outpatient care.

6.1.3. Short-term child mortality

Finally, we examine whether outpatient subsidy affects the most drastic health outcome: mortality. We fully recognize that mortality rate among this age range in Japan is extremely low (in fact, there are only 68 deaths or 0.107% of our sample) and hence we may lack the statistical power to draw precise inferences.⁴³ Nonetheless, we still examine the mortality for the sake of completeness as we believe that the mortality is arguably the most objective health outcome.

Here instead of simple OLS, we account for the interval-censored nature of the mortality data in discrete-time duration model (Jenkins 1995). Specifically, we estimate the following complementary log-log regression model (which is a discrete analog of the continuous proportional hazards model) through maximum-likelihood:

$$Pr(death = 1)_{iat} = \alpha + \sum_{A=7}^{14} \beta_A \{subsidized_{iat} \times 1(Age A)\} + f(a) + \delta_{year} + \theta_{month} + \gamma Female_i + \varepsilon_{iat} - [6]$$

where $Pr(death = 1)_{iat}$ takes one if a child *i* dies at age *a* (in months) in time *t* (in months), and *f*(*a*) captures the underlying baseline hazard. We also control for year FE, month FE as well as a female dummy.⁴⁴

⁴³ The mortality dummy takes one if either 1) enrollment data indicate death as the reason for drop-out from the data, or 2) claims data indicate death as a result of treatment at the medical institutions. We recognize that our data may not capture all the deaths if, for example, children die outside of medical institutions (e.g., home) *and* the death is not reported to the insurers although as many as 83.1% of children aged 5–14 die at hospital (Ministry of Health, Labour and Welfare 2010).

⁴⁴ We cannot include year-month FE and municipality FE since there is no death at every year-month and every municipality. As a result, many observations are dropped from the sample if we include them in the complementary log-log regression.

Appendix Figure H-1 plots the series of β_A when f(a) is log of age in months. While the estimates are somewhat noisy, none of the estimates at any ages are statistically and economically significant. We also experiment with different functional forms of f(a) including linear in age, and age (in year) dummies, which captures the underlying baseline hazard more flexibly. Appendix Table H-1 shows that the estimates are very similar. Thus, at least in the short-run, we find little evidence on reduction in child mortality while we need to interpret this with considerable caution due to very low child mortality rate.⁴⁵

6.2. Evidence of low-value or costly care

Then, we next turn to examine whether we can find clear evidence that subsidy induced care results in low-value or costly care.

6.2.1. Off-hour visits

One concern of generous subsidy is that children (and hence mothers) exploit the opportunity by increasing off-hour visits outside of regular hours because additional fees for off-hour visits are also subsidized. This may place a substantial burden on the workload of the physicians as well as increase medical spending as the fees for off-hour visits are set higher by national fee schedule. On the other hand, if these visits are indeed urgent and not discretionary, the generous subsidy may have little impact on this type of visits. While this issue has been repeatedly raised in the media, there is no formal analysis.⁴⁶

We divide the visits into three categories: regular-hour visits, off-hour visits, and midnight/holiday visits. Under the national fee schedule, additional fees for off-hour visits and midnight/holiday visits are charged on top of the fees for regular-hour visits, and thus from the billing information we know the timing of the outpatient visit within a day.⁴⁷ Appendix Table I-1 provides the list of billing codes for these non regular-hour visits and corresponding additional fees. As a benchmark, fees for regular-hour visits during the sample period are roughly 2.8 and 0.7 thousand JPY (roughly USD28 and 7) for the first visit and revisits, respectively. Appendix Table I-1 shows the *additional* fees charged for off-hour visits—which are typically 0.85 and 0.65 thousand JPY (roughly USD8.5 and 6.5) for the first visit and

⁴⁵ Recent studies find some positive effect of Medicaid introduction (Goodman-Bacon 2016) and expansion on *long-term* health outcomes (Brown *et al.* 2016; Miller and Wherry forthcoming; Wherry *et al.* forthcoming). Note that their studies are likely to find the larger impacts as the focus is on the provision of health insurance (extensive margin), and the targeted population is more disadvantaged.

⁴⁶ Municipalities have indeed concerned that subsidy for child health care may increase unnecessary off-hour visits. See, for example, an article from the leading newspaper in Japan (Nikkei 2017).

⁴⁷ For example, suppose the regular hours of a clinic are registered from 9 am to 5 pm. Then, any visits outside of the regular hours are either off-hour visits or midnight/holiday visits. As the midnight visits are normally defined by visits between 10 pm and 6 am, the visits outside of regular hours but not during midnight—which are between 5 pm to 10 pm and 6 am to 9 am—are considered to be the off-hour visits. Holiday visits are visits in the holiday. We combine midnight and holidays visits as fees for these two types of visits are set higher than other visits by national fee schedule.

revisits, respectively—are relatively high, making these visits costly. Moreover, the additional fees for midnight/holiday visits are set much higher than that of off-hour visits. Note that the medical institutions can charge only one billing code from the list for each visit on top of the fee for regular-hour visit.

Figure 9 plots the estimates (β_A) for regular-hour visits (for references), off-hour visits, and midnight/holiday visits. The corresponding estimates are reported in Table 6. The outcome here is the frequency of visits by each type of visit. Since the majority of the visits are regular-hour visits (89.1% of total visits), Panel A shows the similar pattern as our baseline estimates reported in Figure 5, and the arc-elasticities are stable around -0.1 throughout the age ranges.

Interestingly, Panel B shows that the costly off-hour visits—which account for 8.4% of total visits—also increase due to subsidy. The estimates are slightly increasing in age, and are statistically significant at least above age 9. These results validate the concern that generous subsidy increases burden on the workload of physicians by inducing children to make off-hour visits. The arc-elasticity is also increasing in age, ranging from as low as -0.076 at age 7 to -0.184 at age 14. Importantly, the arc-elasticity for off-hour visits is much larger in magnitude than that of regular-hour visits at older ages. This indicates that at least at the older ages, off-hours visits seem to be more discretionary and less urgent than regular-hour visits, casting some doubt on the generous subsidy for old children. In contrast, we do not see any increases in midnight/holiday visits (which only accounts for 2.5% of total visits) in Panel C. The arc-elasticities are not statistically distinguishable from zero while they are not precisely estimated. These results suggest that the visits at midnight or holidays are indeed very serious cases, and thus children and mothers are less price elastic for these unavoidable visits, which seems plausible. Appendix Figure I-2 and Table I-2 report the estimates on outpatient spending (instead of frequency of visits) and find similar results.⁴⁸

In sum, the subsidy for child health care seems to increase not only the regular-hour visits but also costly off-hour visits, which may increase not only the cost but also the workload of physicians.⁴⁹ From the policy standpoint, the subsidy by the *municipal* government partially undoes the effort of the *national* government to discourage costly off-hour visits for non-serious reasons by setting higher fees for these visits. We did not find evidence, however, that the subsidy increases midnight/holiday visits when the health care resources (e.g., physicians and nurses) are most scarce.

⁴⁸ Note that the spending here only includes consultation fees and does not include any fees related to treatments during the visits.

⁴⁹ It is certainly possible that additional cost of off-hour visits may be partially offset by the opportunity cost of working mothers who may need to leave the work to take children to outpatient care during the regular hours in the absence of subsidy. Ultimately, the availability of free off-hour visits may affect the labor supply of parents. Unfortunately, since our claims data do not include any parental information, we cannot investigate such possibilities.

6.2.2. Inappropriate use of antibiotics

Another concern of generous subsidy for children is that it may increase the inappropriate use of medications. Since more than half of the outpatient spending is occupied by medication and related expenses (54.1%), this is a valid concern worth investigating. Specially, the biggest worry is the use of antibiotics for diagnoses that are not recommended because such inappropriate use leads to both antibiotic resistance and adverse events. For example, the antibiotic-resistant infections annually affect at least 2 million people and 23,000 people dies as a direct result of these infections in the United States (Centers for Disease Control and Prevention 2013). The Japanese government has only recently started addressing misuse of antibiotics by issuing a prescription guideline in 2017.

We follow Fleming-Dutra *et al.* (2016) to create the list of diagnoses for which antibiotics are not recommended.⁵⁰ Appendix Table J-1 presents the list with corresponding ICD10 as well as summary statistics of antibiotic usage. For example, antibiotics use for children with bronchitis and asthma are considered inappropriate. Even without subsidy, roughly 20% of the children diagnosed for any of these diseases are prescribed with antibiotics (Column 5), pointing out the potential misuse of antibiotics for children in Japan. Similarly, the average antibiotics spending conditional on being diagnosed for any of them is 0.24 thousand JPY (Column 6) and the frequency of antibiotics prescriptions is 0.94 per personmonth (Column 7) without subsidy. Both numbers are far from zeros.

It can be problematic if the subsidy increases the number of children in these diagnoses who are prescribed with antibiotics. To investigate this possibility, we estimate the equation [4] where the outcome is the interaction of being diagnosed as any of these diseases and total spending on antibiotics in Panel A, and frequency of antibiotics prescriptions in Panel B. Panel A of Figure 10 (and Table 7) shows that the subsidy increases the spending on antibiotics prescriptions increases by 0.020 thousand JPY, which is 17–38% off the mean. Similarly, the frequency of antibiotics prescriptions increases by 0.039 to 0.070 (20–36% off the mean) in Panel B. Thus, our results suggest that generous subsidy seems to increase the inappropriate use of antibiotics, potentially leading to more antibiotic-resistant infections and adverse effects.

6.3. Price responsiveness by health status

Finally, we examine whether the effect of cost-sharing varies by patient health status.⁵¹ One

⁵⁰ See Appendix Section J for the details on creating the list.

⁵¹ A few papers examine the heterogeneity in price responsiveness by the patient health status. For example, RAND HIE finds no difference between healthier and sicker patients (Manning *et al.* 1987). More recently, Chandra *et al.* (2014) examine the same question among the low-income non-elderly population in Massachusetts. Their findings are rather mixed. Brot-Goldberg *et al.* (2017) documents that the sickest quartile of consumers reduces spending most in response to the introduction of high-deductible health plan in the US, which contrasts our results. Fukushima *et al.*

might expect that as Manning *et al.* (1987) conjectured, medical treatments are less discretionary for sicker patients, and thus sicker patients may be less price responsive than healthier patients. If true, generous subsidy or lack of it would have relatively little effect on sicker patients. If, on the other hand, sicker patients are more price responsive, lack of subsidy may substantially affect the chance of the sick to receive care.

Our longitudinal data allows us to examine history-dependent demand responses. We determine each child's health status by the outpatient spending in the first 6 months since the child is observed at different timing in the claims data. Then, we divide children into two types (i.e., sicker or healthier) by the median spending for each age (in years) and the subsidy status at the first entry to data. Using prior spending as an indicator for health status has been used in previous studies (e.g., Dranove *et al.* 2003). Without the subsidy, the probability of having at least one outpatient visit in a month is 44% for the sick, which is substantially higher than that for the healthy (20%), as expected (see the last row in Table 8). Similarly, the sicker children spend on average 6.89 thousand JPY per month, which is more than three times higher than that of the healthy children (2.04 thousand JPY per month). We estimate the model separately for each group.

Figure 11 (and Table 8) shows that health status does indeed affect children's response to cost sharing. Specifically, the healthier children are much more price sensitive than that of the sicker children. As for an outpatient visit dummy, while the arc-elasticities for the sick range from -0.055 to -0.075, those for the healthy range from -0.12 to -0.16, which is substantially larger in magnitude than that for the sick at any ages. While it is a bit noisier, the same story holds for the outpatient spending. These results shows that sicker children are not the ones who forgo treatments most in the absence of subsidy. Conversely, our results indicate subsidy-induced medical spending for the healthier children are more discretionary and relatively low-value. In Appendix Section K, we experiment with different windows (9, and 12 months) to calculate the patient health status and find qualitatively similar results across the windows.^{52,53}

⁵² We also examine the price elasticities by gender. While the raw outpatient spending is always higher for boys than girls at any ages, the price elasticities are very similar across gender (results available upon request).
⁵³ While we have limited information on supply-side, JMDC data categorizes medical providers into four types: public hospitals, teaching hospitals, other hospitals, and clinics (similar to office visits in the US). In Japan, hospitals are defined as medical institutions with 20 or more beds, and clinics are those with less than 20 beds. There are no forprofit hospitals in Japan in the sense that medical institutions are prohibited from issuing a bond. Since the same national fee is applied to all the medical providers, the sole incentive for patients to visit hospital rather than clinics is because people tend to believe that hospital is of higher quality while the waiting time is longer, and making appointment is harder. Most outpatient visits are at clinics (77.1% of spending and 89.5% of frequency of visits), and thus we combine all the hospital visits into one category. Not surprisingly, we find that almost all the increases in outpatient spending come from clinics (results available upon request). Since people have much easier access to small

⁽²⁰¹⁶⁾ examine the elderly in Japan and find that, similar to ours, the healthier elderly are more price responsive than the sicker elderly.

7. Conclusion

Understanding the price responsiveness to health care is a central question in health economics and the fundamental issue for the optimal design of health insurance. However, past studies on price elasticities are predominantly concentrated on the adults and elderly, and surprisingly little is known about children. In this paper, we examine the effect of patient cost-sharing on health care utilization among children by exploiting more than 5000 regional and over-time variations on subsidy availability. Importantly, the eligibility for subsidy is always tied to the age of children, allowing us to estimate the price elasticity of children at each age from 7 to 14.

We find that the reduction in cost-sharing from 30% (national level) to 0% (free) increases the outpatient spending by 22–31%. The arc-elasticity is relatively stable around –0.1 across the age ranges we examine. While the considerable caution is needed to compare the elasticities estimated across countries or time periods, the arc-elasticity for children estimated in this paper is somewhat smaller than that of RAND HIE for non-elderly in the US, and Shigeoka (2014) and Fukushima *et al.* (2016) for the elderly in Japan.

We further examine the utilization patterns from various dimensions to understand whether changes in utilization largely reflect beneficial or low-value care. We show that the increases in outpatient visits do not translate to clear benefits in the form of reduction in hospitalization by "avoidable" conditions nor reduction in short-run mortality. We also document that the subsidy has some negative effects by increasing inappropriate use of antibiotics and costly off-hour visits. Furthermore, the healthier children—measured by the prior spending—is much more price sensitive than the sickly children, which appears to indicate that the subsidy induces discretionary health care utilizations. Taken individually, each piece of empirical evidence may not be sufficient to establish the existence of wasteful utilization. But taken together, the weight of the evidence supports the notion that the aggressive expansion of child health care subsidy may lead to the increases of the low-value and costly outpatient visits.

This paper is subject to a couple of limitations. The biggest limitation is that we cannot investigate both short-term as well as long-term health outcomes except for short-run mortality. Since health is stock, the better access to preventive care during childhood may translate into the improvement in the long-run health, which can potentially justify the generous subsidy for child health care. Another important limitation is that our insurance claims data do not include basic parental characteristics such as income and education. This may be especially important in case of young children as their decision making is heavily influenced by mothers. To the best of our knowledge, such monthly data with age,

clinics rather than large hospitals, these visits are probably less serious and more discretionary, providing another suggestive evidence of increases in non-essential care.

municipality of residence, health care utilization, and any household characteristics, do not exist in Japan—due mainly to the lack of individual identifier in Japan such as social security number in the US—but this definitely leaves an avenue for future research.

References

- Aron-Dine, Aviva, Liran Einav, Amy Finkelstein, and Mark Cullen. (2015) "Moral Hazard in Health Insurance: Do Dynamic Incentives Matter?" *Review of Economics and Statistics* 97(4): 725–741.
- **Baicker, Katherine, and Dana Goldman.** (2011) "Patient Cost-Sharing and Healthcare Spending Growth" *Journal of Economic Perspectives* 25(2): 47–68.
- **Baicker, Katherine, Sendhil Mullainathan, and Joshua Schwartzstein.** (2015) "Behavioral Hazard in Health Insurance." *Quarterly Journal of Economics* 130(4): 1623–1667.
- **Bessho, Shunichiro.** (2012) "Medical care subsidy for, hospital visits of, and health status of children." *The Quarterly of Social Security Research* (in Japanese) 47(4): 413–430.
- Brot-Goldberg, Zarek C., Amitabh Chandra, Benjamin R. Handel, and Jonathan T. Kolstad. (2017) "What does a Deductible Do? The Impact of Cost-Sharing on Health Care Prices, Quantities, and Spending Dynamics." *Quarterly Journal of Economics* 132(3): 1261–1318.
- Brown, David W., Amanda E. Kowalski, and Ithai Z. Lurie. (2016) "Medicaid as an Investment in Children: What is the Long-Term Impact on Tax Receipts?" *NBER Working Paper 20835*.
- **Cabral, Marika.** (2017) "Claim Timing and Ex Post Adverse Selection." *Review of Economic Studies* 84: 1–44.
- **Case, Anne, Angela Fertig, and Christina Paxson.** (2005) "The Lasting Impact of Childhood Health and Circumstance." *Journal of Health Economics* 24(2): 365–389.
- **Centers for Disease Control and Prevention.** (2013) "Antibiotic resistance threats in the United States, 2013." <u>http://www.cdc.gov/drugresistance/threat-report-2013/</u> (Last accessed on October 26, 2017).
- **Chandra, Amitabh, Jonathan Gruber, and Robin McKnight.** (2010) "Patient Cost-Sharing and Hospitalization Offsets in the Elderly." *American Economic Review* 100(1): 193–213.
- **Chandra, Amitabh, Jonathan Gruber, and Robin McKnight.** (2014) "The impact of patient costsharing on low-income populations: Evidence from Massachusetts." *Journal of Health Economics* 33: 57–66.
- **Currie, Janet.** (2009). "Healthy, Wealthy, and Wise: Socioeconomic Status, Poor Health in Childhood, and Human Capital Development." *Journal of Economic Literature* 47(1): 87–122.
- Currie, Janet, and Jonathan Gruber. (1996) "Health insurance eligibility, utilization of medical care, and child health." *Quarterly Journal of Economics* 111(2): 431–466.
- Currie, Janet, Lucas Davis, Michael Greenstone, and Reed Walker. (2015) "Environmental Health Risks and Housing Values: Evidence from 1,600 Toxic Plant Openings and Closings." *American Economic Review*, 105(2): 678–709.
- Cutler, David M. (1995) "The Incidence of Adverse Medical Outcomes Under Prospective Payment." *Econometrica* 63(1): 29–50.
- **Dafny, Leemore S., and Jonathan Gruber.** (2005) "Public insurance and child hospitalizations: access and efficiency effects." *Journal of Public Economics* 89(1): 109–129.
- **Dranove, David, Daniel Kessler, Mark McClellan, and Mark Satterthwaite.** (2003) "Is More Information Better? The Effects of `Report Cards' on Health Care Providers." *Journal of Political Economy* 111(3): 555–588.

- Ellis, Randall P. (1986) "Rational Behavior in the Presence of Coverage Ceilings and Deductibles." *RAND Journal of Economics* 17(2): 158–175.
- Finkelstein, Amy, Sarah Taubman, Bill Wright, Mira Bernstein, Jonathan Gruber, Joseph P. Newhouse, Heidi Allen, and Katherine Baicker. (2012) "The Oregon Health Insurance Experiment: Evidence from the First Year." *Quarterly Journal of Economics* 127(3): 1057–1106.
- Fleming-Dutra *et al.* (2016) "Prevalence of inappropriate antibiotic prescriptions among US ambulatory care visits, 2010-2011." *JAMA* 315(17): 1864–1873.
- **Foxman** *et al.* (1987) "The effect of cost sharing on the use of antibiotics in ambulatory care: Results from a population-based randomized controlled trial." *Journal of Chronic Diseases* 40(5): 429–437.
- **Fukushima, Kazuya, Sou Mizuoka, Shunsuke Yamamoto, and Toshiaki Iizuka.** (2016) "Patient cost sharing and medical expenditures for the Elderly." *Journal of Health Economics* 45: 115–130.
- Gadomski, Anne, Paul Jenkins, and Melissa Nichols. (1998) "Impact of a Medicaid primary care provider and preventive care on pediatric hospitalization." *Pediatrics* 101(3).
- Goodman-Bacon, Andrew. (2016) "The Long-Run Effects of Childhood Insurance Coverage: Medicaid Implementation, Adult Health, and Labor Market Outcomes." *NBER Working Paper No. 22899*.
- Goodman-Bacon, Andrew. (2018) "Public Insurance and Mortality: Evidence from Medicaid Implementation." *Journal of Political Economy* 126(1): 216–262.
- Han, Hsing-Wen, Hsien-Ming Lien, and Tzu-Ting Yang. (2016) "Patient Cost Sharing and Healthcare Utilization in Early Childhood: Evidence from a Regression Discontinuity Design." Unpublished manuscript.
- Hossein, Zare, and Anderson Gerard. (2013) "Trends in cost sharing among selected high income countries—2000–2010." *Health Policy* 112(1–2): 35–44.
- **Iizuka, Toshiaki.** (2007) "Experts' Agency Problems: Evidence from the Prescription Drug Market in Japan." *RAND Journal of Economics* 38(3): 844–862.
- **Iizuka, Toshiaki.** (2012) "Physician Agency and Adoption of Generic Pharmaceuticals." *American Economic Review* 102(6): 2826–2858.
- Ikegami, Naoki, and John C. Campbell. (1995) "Medical Care in Japan." New England Journal of Medicine 333: 1295–1299.
- Japan Federation of Democratic Medical Institutions. (2015) <u>https://www.min-iren.gr.jp/hokoku/data/hokoku_h27dpc/160429_48.pdf</u> (Last accessed on October 26, 2017).
- Jenkins, Stephen P. (1995) "Easy estimation methods for discrete time duration models." Oxford Bulletin of Economics and Statistics 57(1): 129–136.
- Kaestner, Robert, and Anthony T. Lo Sasso. (2015) "Does Seeing the Doctor More Often Keep You Out of the Hospital?" *Journal of Health Economics* 39: 259–272.
- Kaestner, Robert, Theodore J. Joyce, and Andrew D. Racine. (2001) "Medicaid eligibility and the incidence of ambulatory care sensitive hospitalizations for children." *Social Science & Medicine* 52(2): 305–313.
- Keeler, Emmett B., Joseph P. Newhouse, and C. E. Phelps. (1977) "Deductibles and the Demand for Medical Care Services: The Theory of a Consumer Facing a Variable Price Schedule under Uncertainty." *Econometrica* 45(3): 641–655.
- Keeler, Emmett B., and John E. Rolph. (1988) "The Demand for Episodes of Treatment in the Health Insurance Experiment." *Journal of Health Economics* 7 (4): 337–367.
- Kondo, Ayako, and Hitoshi Shigeoka. (2013) "Effects of Universal Health Insurance on Health Care Utilization and Supply-Side Responses: Evidence from Japan." *Journal of Public Economics* 99: 1–23.

- Lee, David, and Frank Levy. (2012) "The Sharp Slowdown in Growth of Medical Imaging: An Early Analysis Suggests Combination of Policies Was the Cause." *Health Affairs* 31(8): 1–9.
- Leibowitz, Arleen, Willard G. Manning, Jr, Emmett B. Keeler, Naihua Duan, Kathleen N. Lohr, Joseph P. Newhouse. (1985) "Effect of Cost-Sharing on the Use of Medical Services by Children: Interim Results from a Randomized Controlled Trial." *Pediatrics* 75(5).
- Manning, Willard G., Joseph P. Newhouse, Naihua Duan, Emmett B. Keeler, and Arleen Leibowitz. (1987) "Health Insurance and the Demand for Medical Care: Evidence from a Randomized Experiment." *American Economic Review* 77(3): 251–277.
- Miller, Sarah and Laura R. Wherry. (2018) "The Long-Term Effects of Early Life Medicaid Coverage." *Journal of Human Resources*, forthcoming.
- Ministry of Health, Labour and Welfare. (2010) Vital statistics. <u>https://www.e-stat.go.jp/stat-search/files?page=1&layout=datalist&toukei=00450011&tstat=000001028897&cycle=7&year=20100&month=0&tclass1=000001053058&tclass2=000001053061&tclass3=000001053065&stat_infid=000011816828&result_back=1&second2=1 (last accessed on November 22, 2017)</u>
- **Ministry of Health, Labour and Welfare.** (2014) "Overview of Japan's Health Care System" (<u>http://www.mhlw.go.jp/file/06-Seisakujouhou-12400000-Hokenkyoku/0000172084.pdf</u> (last accessed on March 1, 2018)
- Newhouse, Joseph P., and the Insurance Experiment Group. (1993) Free for All: Lessons from the RAND Health Insurance Experiment. Cambridge, MA: Harvard University Press.
- Nikkei. (2017) <u>https://www.nikkei.com/article/DGXMZO21776910S7A001C1L61000/</u> (in Japanese) (last accessed on February 14, 2018).
- Nilsson, Anton, and Alexander Paul. (2015) "The Effect of Copayments on Children's and Adolescents' Use of Medical Care." *Unpublished manuscript*.
- Shampanier, Kristina, Nina Mazar, and Dan Ariely. (2007) "Zero as a Special Price: The True Value of Free Products." *Marketing Science* 26(6): 742–757.
- Shigeoka, Hitoshi. (2014) "The Effect of Patient Cost-sharing on Utilization, Health and Risk Protection." *American Economic Review* 104(7): 2152–2184.
- Shigeoka, Hitoshi. (2015) "School entry cutoff date and the timing of births." *NBER Working Paper* 21402.
- Smith, James P. (2009) "The Impact of Childhood Health on Adult Labor Market Outcomes." *Review* of Economics and Statistics 91(3): 478–489.
- Statistics Bureau. (2015) Population Census. <u>http://www.e-stat.go.jp/SG1/estat/GL38020101.do?_toGL38020101_&tstatCode=000001080615</u> (last accessed in November 22, 2017)
- **Trivedi, Amal N., Husein Moloo, and Vincent Mor.** (2010) "Increased Ambulatory Care Copayments and Hospitalizations among the Elderly." *New England Journal of Medicine* 362(4): 320–328.
- Wherry, Laura R., Sarah Miller, Robert Kaestner, and Bruce D. Meyer. (2017) "Childhood Medicaid Coverage and Later Life Health Care Utilization." *Review of Economics and Statistics* forthcoming.
- **Zhong, Hai.** (2011) "Effect of patient reimbursement method on health-care utilization: Evidence from China." *Health Economics* 20(11): 1312–1329.

Figure 1: Time series of maximum age fully covered by child care subsidy



Notes: The data is unbalanced monthly panel where the unit of observation is municipality. There are total of 165 municipalities. Note that this figure reflects the compositional changes of municipalities as the number of municipality increases at the later period in our claim data. Importantly, within the municipalities, the subsidy expansion is always monotonic—that is, there are no single municipality that *lowers* the maximum age during this period (April 2005–March 2015). The spike in April 2008 is explained by the fact that the central government has expanded the eligibility age for national-level subsidy (i.e., 20% coinsurance rate) from age 3 to 6 (till the start of primary school). While Figure 1 clearly shows that all municipalities in our sample have already provided the subsidy till age 6 by April 2008, this national-level subsidy expansion eases the budgetary burden on municipalities as the part of the cost is now covered by the central government. For this reason, we see the highest number of municipality-level subsidy expansions in April 2008 to ages higher than age 6 (See Appendix Figure A-1 on the precise timing of all policy changes).

Figure 2: Age profiles of utilizations by subsidy status

A. Outpatient care (monthly)



B. Inpatient care (yearly)



Notes: Panel A plots the monthly mean of outpatient outcomes, and Panel B plots the yearly mean of inpatient outcomes as inpatient admission is a rare event. An outpatient visit dummy takes one if there is at least one outpatient visit per month, and an inpatient admission dummy takes one if there is at least one hospitalization per month (×1000). The outpatient spending is the monthly spending on outpatient care and the inpatient spending is monthly spending on inpatient care, both of which are measured in thousand JPY (roughly USD10). The dotted lines are age profiles of utilization without subsidy (30% coinsurance, labeled "no subsidy"), and the solid lines are age profiles of utilization with subsidy (0% coinsurance, labeled "subsidized").


Notes: The solid line is the price schedule for cohort born in July 1998 ("younger" cohort) while that of dotted line is the price schedule for cohort born in June 1998 ("older" cohort"), a month before the previous cohort. In this example, the subsidy expansion from age 6 (before the start of primary school) to 15 (end of junior high school) occurs in October 2007.

Figure 4: Event study

A. Outpatient visit dummy



B. Outpatient spending (in 1K JPY)



Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). "Better" indicates the subsidy expansion which lowers the price of health care from 30% to 0%, and "Worse" indicates subsidy expiration that raises the price from 0% to 30%. The solid lines plot the estimates from a variant of estimation equation [4] where the subsidized dummy is replaced by a series of dummy for each month ranging from 12 months prior to the change in subsidy status to 12 months after the change in subsidy status (T=-12 to +11, where T=0 is the change). The dotted lines are the 95th confidence intervals where standard errors clustered at municipality level are used to construct them. The reference month is 3 months before the subsidy changes (T=-3). The observations within two months from the subsidy changes of the opposite direction are excluded from the sample to account for anticipatory utilization. Note that the scales of *y*-axis are set the same within the panels so that two figures for opposite directions of subsidy changes are visually comparable.

Figure 5: Basic results



Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). The dotted lines are the 95th confidence intervals. The standard errors clustered at municipality level are used for estimates, and the bootstrapped standard errors clustered at municipality with 200 repetitions are used for the arc-elasticities. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Note that the scales of *y*-axis on the arc-elasticities are set the same so that two elasticities are visually comparable.



Figure 6: Asymmetric responses

Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). "Better" indicates the subsidy expansion which lowers the price of health care from 30% to 0%, and "Worse" indicates subsidy expiration that raises the price from 0% to 30%. The dotted lines are the 95th confidence intervals. The standard errors clustered at municipality level are used for estimates, and the bootstrapped standard errors clustered at municipality with 200 repetitions are used for the semi point-elasticity. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Note that the scales of *y*-axis on the semi point-elasticity are set the same so that two elasticities are visually comparable.

Figure 7: Ambulatory Care Sensitive Conditions (ACSC)



Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and an inpatient admission dummy takes one if there is at least one hospitalization per month (\times 1000). See Online Appendix G for the list of ACSC and summary statistics. The dotted lines are the 95th confidence intervals and the standard errors clustered at municipality level are used to construct them. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. ENT stands for Ear, Nose, and Throat. The corresponding arc-elasticities are presented in Online Appendix G.

Figure 8: Offset effects



A. Inpatient admission dummy (×1000)

Notes: An inpatient admission dummy takes one if there is at least one hospitalization per month ($\times 1000$), and inpatient spending is the monthly spending on inpatient care measured in thousand JPY (roughly USD10). The dotted lines are the 95th confidence intervals derived from standard errors clustered at municipality level. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization.

Age

Figure 9: By time of visits

Outcome: Frequency of outpatient visits

A. Regular-hour visits

B. Off-hour visits





Notes: The frequency of outpatient visits is the number of outpatient visits per month. See Online Appendix H which provides the list of billing codes for off-hour and midnight/holiday and corresponding fees that are additionally charged on top of fees for regular-hour visits. The dotted lines are the 95th confidence intervals. The standard errors clustered at municipality level are used for estimates, and the bootstrapped standard errors clustered at municipality with 200 repetitions are used for the arc-elasticities.



Figure 10: Inappropriate use of antibiotics

Notes: The outcome is monthly outpatient spending on antibiotics considered as inappropriate measured in thousand JPY (roughly USD10) in Panel A and the number of prescriptions for the antibiotic per month in Panel B. See Online Appendix J for the list of ICD10 codes and the summary statistics for inappropriate use of antibiotics. The dotted lines are the 95th confidence intervals. The standard errors clustered at municipality level are used for estimates, and the bootstrapped standard errors clustered at municipality with 200 repetitions are used for the arc-elasticities. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization.



Figure 11: Price responsiveness by health status

Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). We determine each child's health status by the outpatient spending in the first 6 months since the child is observed in the claim data. Then, we divide children into two types (i.e., sicker or healthier) by the median spending for each age (in years) and the subsidy status at the first entry to data. See the main text for details. The standard errors clustered at municipality level are used for estimates, and the bootstrapped standard errors clustered at municipality with 200 repetitions are used for the arc-elasticities. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Note that the scales of *y*-axis on the arc-elasticities are set the same so that two elasticities are visually comparable.

Variable	Mean	SD	Min	Max
A. Municipality (N=165)				
Average length observed (months)	76.59	32.77	5	120
Subsidy info				
Number of policy changes	1.20	1.12	0	5
At least one policy change	68.5%	0.47	0	1
B. Individual (N=63,590)				
Average length observed (months)	36.22	31.14	2	119
Subsidy info				
Number of subsidy status changes	0.39	0.80	0	5
At least one subsidy status change	21.8%	0.41	0	1
At least one subsidy expansion ("better")	16.5%	0.37	0	1
At least one subsidy expiration ("worse")	19.3%	0.39	0	1
Characteristics				
Female	48.8%	0.50	0	1
Age (in years)	10.86	2.85	6.08	15.92
C. Overall (N= 2,303,335 person-month)				
Subsidy info				
Subsidized	71.0%	0.45	0	1
In-kind (when subsidized)	99.9%	0.03	0	1
Income restriction (when subsidized)	1.5%	0.12	0	1
Utilization				
Outpatient visit dummy	40.7%	0.49	0	1
Outpatient spending (in 1K JPY)	6.09	25.33	0	9,336
Outpatient spending (outpatient visit >0)	14.95	37.98	0.08	9,336
N of outpatient visits	0.83	1.46	0	34
N of outpatient visits (outpatient visit >0)	2.04	1.65	0	34
Inpatient admission dummy	0.28%	0.05	0	1
Inpatient spending (in 1K JPY)	1.15	35.24	0	6,084
Inpatient spending (inpatient admission >0)	406.52	523.57	5.23	6,084

Table 1: Summary statistics

Notes: The outpatient and inpatient spending are measured in thousands JPY (roughly 10USD).

-	A.	Outpatier	nt visit dumm	y	B. Ou	itpatient s	pending (in 1K	JPY)
	(1)	(2)		(3)	(4)	
	Estimate	(SE)	Arc- elasticities	[SE]	Estimate	(SE)	Arc- elasticities	[SE]
Subsidized $ imes$								
Age7	0.075^{***}	(0.006)	-0.083***	[0.005]	1.376***	(0.236)	-0.111***	[0.008]
Age8	0.068^{***}	(0.006)	-0.078***	[0.004]	1.278***	(0.179)	-0.107***	[0.009]
Age9	0.070^{***}	(0.005)	-0.086***	[0.004]	1.131***	(0.171)	-0.099***	[0.009]
Age10	0.073***	(0.005)	-0.094***	[0.004]	1.350***	(0.221)	-0.125***	[0.009]
Age11	0.066***	(0.006)	-0.090***	[0.004]	1.113***	(0.211)	-0.106***	[0.009]
Age12	0.064***	(0.004)	-0.092***	[0.004]	0.872***	(0.223)	-0.082***	[0.010]
Age13	0.062***	(0.004)	-0.092***	[0.004]	0.811***	(0.218)	-0.076***	[0.009]
Age14	0.060^{***}	(0.004)	-0.092***	[0.004]	0.998***	(0.134)	-0.095***	[0.010]
In-kind	0.047^{***}	(0.014)			0.440	(0.388)		
Income restriction	-0.020**	(0.009)			-0.561	(0.372)		
R-squared	0.2	23			0.5	51		
N	2,205	,647			2,205	,647		
N of Individual	63,5	30			63,5	30		
Mean wo subsidy	0.3	2			4.4	.9		

 Table 2: Basic results

Notes: An outpatient visit dummy takes one if an individual makes at least one outpatient visit per month and zero otherwise. The outpatient spending is total monthly spending on outpatient care measured in thousands JPY (roughly 10USD). All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. For the estimates, the standard errors clustered at the municipality level are reported in parenthesis. For the arc-elasticities, the bootstrapped standard errors with 200 repetitions clustered at municipality level are reported in brackets. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Significance levels: *** p<0.01, ** p<0.05, * p<0.10

	Α	. Outpatie	nt visit dumm	ıy	B. Ou	tpatient sp	ending (in 1K	JPY)
	(1))	(2))	(3))	(4)
	Estimate	(SE)	Semi point- elasticity	[SE]	Estimate	(SE)	Semi point- elasticity	[SE]
Subsidized $ imes$								
Age7 × Better	0.060^{***}	(0.009)	-0.485***	[0.053]	1.064^{***}	(0.291)	-0.691***	[0.112]
Age8 × Better	0.069^{***}	(0.008)	-0.590***	[0.048]	1.106***	(0.278)	-0.759***	[0.109]
Age9 × Better	0.060^{***}	(0.006)	-0.562***	[0.050]	0.925***	(0.244)	-0.691***	[0.105]
Age10 × Better	0.067^{***}	(0.007)	-0.659***	[0.042]	1.056***	(0.283)	-0.836***	[0.109]
Age11 × Better	0.072^{***}	(0.007)	-0.748***	[0.038]	1.135***	(0.257)	-0.880***	[0.102]
Age12 \times Better	0.069***	(0.006)	-0.763***	[0.041]	0.936***	(0.269)	-0.700***	[0.092]
Age13 \times Better	0.066***	(0.005)	-0.751***	[0.043]	0.809^{**}	(0.325)	-0.582***	[0.091]
Age14 \times Better	0.067^{***}	(0.006)	-0.775***	[0.040]	0.724**	(0.315)	-0.524***	[0.094]
Subsidized $ imes$								
Age7 × Worse	-0.070***	(0.008)	-0.462***	[0.033]	-1.633***	(0.333)	-0.746***	[0.062]
Age8 × Worse	-0.068***	(0.008)	-0.476***	[0.039]	-1.633***	(0.340)	-0.769***	[0.082]
Age9 × Worse	-0.066***	(0.009)	-0.483***	[0.049]	-1.493***	(0.326)	-0.723***	[0.079]
Age10 \times Worse	-0.062***	(0.008)	-0.471***	[0.042]	-1.601***	(0.274)	-0.813***	[0.082]
Age11 × Worse	-0.066***	(0.007)	-0.538***	[0.042]	-1.551***	(0.222)	-0.837***	[0.081]
Age12 \times Worse	-0.061***	(0.007)	-0.522***	[0.042]	-1.416***	(0.274)	-0.774***	[0.080]
Age13 \times Worse	-0.058***	(0.005)	-0.519***	[0.044]	-1.157***	(0.214)	-0.639***	[0.085]
Age14 \times Worse	-0.061***	(0.005)	-0.560***	[0.045]	-1.155***	(0.200)	-0.657***	[0.084]
In-kind	0.050^{***}	(0.014)			0.452	(0.395)		
Income restriction	-0.020**	(0.009)			-0.731**	(0.346)		
R-squared	0.2	3			0.5	1		
N	2,144,	,756			2,144	,756		
N of Individual Mean wo subsidy	62,6 0 3	09 2			62,6 4 4	09 9		

Table 3: Asymmetric responses

Notes: An outpatient visit dummy takes one if an individual makes at least one outpatient visit per month and zero otherwise. The outpatient spending is total monthly spending on outpatient care measured in thousands JPY (roughly 10USD). "Better" indicates the subsidy expansion which lowers the price of health care from 30% to 0%, and "Worse" indicates subsidy expiration that raises the price from 0% to 30%. All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. For the estimates, the standard errors clustered at the municipality level are reported in parenthesis. For the semi point-elasticity, the bootstrapped standard errors with 200 repetitions clustered at municipality level are reported in brackets. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Significance levels: *** p<0.01, ** p<0.05, * p<0.10

		A. Any	y ACSC			B. 1	ENT			C. As	sthma	
	(1)	(2)	(3)	(4)	(5)	(6)
	An <i>out</i> dum	p <i>atient</i> Imy	An <i>inp</i> dummy	atient (×1000)	An <i>outp</i> dum	<i>patient</i> my	An <i>inp</i> dummy	atient (×1000)	An <i>outp</i> dum	<i>patient</i> my	An <i>inp</i> dummy	atient (×1000)
	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)
Subsidized $ imes$												
Age7	0.038***	(0.005)	0.159***	(0.053)	0.027^{***}	(0.004)	0.060^{**}	(0.028)	0.022^{***}	(0.003)	0.089^{***}	(0.018)
Age8	0.034***	(0.005)	0.070	(0.048)	0.024***	(0.004)	0.015	(0.026)	0.020^{***}	(0.003)	0.056^{***}	(0.016)
Age9	0.031***	(0.005)	0.077^{*}	(0.044)	0.023***	(0.004)	0.030	(0.020)	0.015***	(0.002)	0.045***	(0.014)
Age10	0.032***	(0.004)	0.051	(0.053)	0.024***	(0.004)	0.025	(0.020)	0.015***	(0.002)	0.038***	(0.012)
Age11	0.026***	(0.003)	0.033	(0.042)	0.020***	(0.003)	0.030	(0.022)	0.010***	(0.002)	0.025	(0.016)
Age12	0.024***	(0.003)	-0.002	(0.072)	0.017^{***}	(0.003)	0.035	(0.024)	0.009^{***}	(0.002)	0.035**	(0.015)
Age13	0.022***	(0.003)	0.056	(0.044)	0.015***	(0.003)	0.054***	(0.020)	0.009^{***}	(0.002)	0.047***	(0.016)
Age14	0.022***	(0.003)	0.026	(0.043)	0.016***	(0.002)	0.043**	(0.019)	0.008^{***}	(0.002)	0.038***	(0.013)
In-kind	0.030**	(0.015)	0.180^{*}	(0.095)	0.021***	(0.005)	0.053***	(0.017)	0.010	(0.015)	0.058^{**}	(0.025)
Income restriction	-0.008	(0.008)	-0.185***	(0.059)	-0.011	(0.007)	-0.060**	(0.027)	0.000	(0.004)	-0.079**	(0.035)
R-squared	0.2	24	0.1	.4	0.1	.6	0.0)6	0.3	5	0.1	.6
N	2,205	,647	2,205	,647	2,205	,647	2,205	,647	2,205	,647	2,205	,647
N of Individual	63,5	530	63,5	530	63,5	530	63,5	530	63,5	530	63,5	530
Mean wo subsidy	0.1	1	1.0)4	0.0)8	0.3	32	0.0)4	0.2	25

 Table 4: Ambulatory Care Sensitive Conditions (ACSC)

Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and an inpatient admission dummy takes one if there is at least one hospitalization per month (\times 1000). See Online Appendix G for the list of ACSC and summary statistics. All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. For even-numbered columns, we also control for a dummy that takes one if the municipality also offers subsidy for inpatient care. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. ENT stands for Ear, Nose, and Throat. Significance levels: *** p<0.01, ** p<0.05, * p<0.10

	<i>Inpat</i> dummy (<i>ient</i> (×1000)	<i>Inpatient</i> spendin (in 1K JPY) (2)		
	(1))			
	Estimate	(SE)	Estimate	(SE)	
Subsidized $ imes$					
Age7	2.365**	(1.026)	0.814	(0.599)	
Age8	0.958	(0.879)	0.761	(0.618)	
Age9	1.023	(0.918)	0.147	(0.564)	
Age10	0.721	(0.812)	0.009	(0.510)	
Age11	0.104	(1.191)	-0.151	(0.792)	
Age12	0.160	(1.001)	-0.094	(0.692)	
Age13	0.974	(0.730)	0.098	(0.481)	
Age14	0.193	(0.857)	0.116	(0.497)	
In-kind	2.984***	(1.109)	0.316	(0.539)	
Income restriction	-2.423***	(0.757)	-1.306***	(0.453)	
R-squared	0.1	2	0.2	0	
N	2,205	,647	2,205	,647	
N of Individual	63,5	30	63,530		
Mean	2.4	1	1.0	1.00	

 Table 5: Offset effects

Notes: An inpatient admission dummy takes one if there is at least one hospitalization per month (×1000), and inpatient spending is the monthly spending on inpatient care measured in thousand JPY (roughly USD10). All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. We also control for a dummy that takes one if the municipality also offers subsidy for inpatient care. The standard errors clustered at the municipality level are reported in parenthesis. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Significance levels: *** p<0.01, ** p<0.05, * p<0.10

Table 6: By time of visits

Outcome:	Frequenc	y of out	patient	visits
		1		

	A	A. Regula	r-hour visits			B. Off-h	our visits		C. 1	Midnight	/Holiday vis	its
	(1)	(2))	(3)	(4))	(5)	(6)
	Estimate	(SE)	Arc- elasticity	[SE]	Estimate	(SE)	Arc- elasticity	[SE]	Estimate	(SE)	Arc- elasticity	[SE]
Subsidized $ imes$												
Age7	0.169***	(0.028)	-0.115***	[0.007]	0.008	(0.006)	-0.076***	[0.017]	0.001	(0.001)	-0.054*	[0.030]
Age8	0.148^{***}	(0.030)	-0.108***	[0.006]	0.004	(0.006)	-0.043**	[0.017]	0.001	(0.001)	-0.028	[0.033]
Age9	0.115***	(0.027)	-0.091***	[0.007]	0.008^{*}	(0.005)	-0.081***	[0.019]	-0.001	(0.001)	0.030	[0.037]
Age10	0.110***	(0.031)	-0.092***	[0.007]	0.013***	(0.004)	-0.123***	[0.018]	-0.000	(0.001)	0.007	[0.036]
Age11	0.096***	(0.026)	-0.088***	[0.008]	0.017^{***}	(0.004)	-0.163***	[0.019]	-0.000	(0.001)	0.017	[0.039]
Age12	0.079***	(0.023)	-0.078***	[0.007]	0.016***	(0.003)	-0.146***	[0.018]	-0.001	(0.001)	0.065^{*}	[0.035]
Age13	0.082^{***}	(0.020)	-0.085***	[0.007]	0.024***	(0.004)	-0.199***	[0.017]	-0.000	(0.001)	0.015	[0.037]
Age14	0.096***	(0.018)	-0.104***	[0.007]	0.021***	(0.004)	-0.184***	[0.018]	0.000	(0.001)	-0.029	[0.040]
In-kind	0.035	(0.053)			0.031	(0.019)			-0.001	(0.004)		
Income restriction	-0.051	(0.032)			-0.005	(0.010)			-0.001	(0.002)		
R-squared	0.2	28			0.1	4			0.0)4		
Ν	2,205	,647			2,205	,647			2,205	,647		
N of Individual	63,5	530			63,5	530			63,5	30		
Mean wo subsidy	0.4	01			0.0	23			0.0	06		
Share	89.1	1%			8.4	%			2.5	%		

Notes: The frequency of outpatient visits is the number of outpatient visits per month. See Online Appendix I for the list of billing codes for the urgent visits (off-hour visits and midnight/holiday visits) and the corresponding fees that are additionally charged on top of fees for regular-hour visits. All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. For the estimates, the standard errors clustered at the municipality level are reported in parenthesis. For the semi point-elasticity, the bootstrapped standard errors with 200 repetitions clustered at municipality level are reported in brackets. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Significance levels: *** p<0.01, ** p<0.05, * p<0.10

	A. Out	p atient spo (in	ending on ant 1K JPY)	ibiotics	B. Freq	uency of a	ntibiotics pres	criptions
	(1)	(2)		(3)	(4)	
	Estimate	(SE)	Arc- elasticities	[SE]	Estimate	(SE)	Arc- elasticities	[SE]
Subsidized $ imes$								
Age7	0.013***	(0.005)	-0.078***	[0.020]	0.039***	(0.014)	-0.071***	[0.019]
Age8	0.015^{***}	(0.003)	-0.102***	[0.021]	0.051***	(0.012)	-0.104***	[0.019]
Age9	0.014^{***}	(0.003)	-0.108***	[0.022]	0.054***	(0.013)	-0.119***	[0.020]
Age10	0.020^{***}	(0.004)	-0.182***	[0.026]	0.070^{***}	(0.015)	-0.166***	[0.021]
Age11	0.016^{***}	(0.003)	-0.174***	[0.028]	0.063***	(0.009)	-0.163***	[0.022]
Age12	0.016^{***}	(0.003)	-0.205***	[0.030]	0.066***	(0.011)	-0.194***	[0.022]
Age13	0.011***	(0.003)	-0.157***	[0.028]	0.045***	(0.010)	-0.139***	[0.022]
Age14	0.009^{***}	(0.002)	-0.134***	[0.026]	0.042***	(0.006)	-0.139***	[0.022]
In-kind	0.018^{**}	(0.009)			0.087^*	(0.052)		
Income restriction	-0.006	(0.005)			-0.029*	(0.017)		
R-squared	0.0)8			0.0	19		
N	2,205	,647			2,205	,647		
N of Individual	63,5	30			63,5	30		
Mean wo subsidy	0.03	52			0.1	93		

Table 7:	Inappro	priate use	of a	ntibiotics
----------	---------	------------	------	------------

Notes: The outcome is monthly outpatient spending on antibiotics considered as inappropriate in Panel A and the number of prescriptions for the antibiotic per month in Panel B. See Online Appendix J for the list of ICD10 codes and the summary statistics for inappropriate use of antibiotics. All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. For the estimates, the standard errors clustered at the municipality level are reported in parenthesis. For the arc-elasticities, the bootstrapped standard errors with 200 repetitions clustered at municipality level are reported in brackets. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Significance levels: *** p<0.01, ** p<0.05, * p<0.10

	-	He	althier	-		S	bicker	
	(1)	(2)		(3)	(4)	
	Estimate	(SE)	Arc- elasticities	[SE]	Estimate	(SE)	Arc- elasticities	[SE]
Subsidized $ imes$								
Age7	0.085^{***}	(0.008)	-0.136***	[0.009]	0.081^{***}	(0.011)	-0.067***	[0.005]
Age8	0.084^{***}	(0.008)	-0.140***	[0.008]	0.062***	(0.008)	-0.055***	[0.012]
Age9	0.084^{***}	(0.006)	-0.150***	[0.008]	0.067^{***}	(0.008)	-0.063***	[0.005]
Age10	0.086^{***}	(0.006)	-0.160***	[0.009]	0.068^{***}	(0.006)	-0.067***	[0.005]
Age11	0.077^{***}	(0.007)	-0.152***	[0.009]	0.057***	(0.008)	-0.060***	[0.006]
Age12	0.069***	(0.005)	-0.143***	[0.009]	0.060^{***}	(0.005)	-0.066***	[0.005]
Age13	0.058^{***}	(0.006)	-0.123***	[0.008]	0.065***	(0.004)	-0.075***	[0.005]
Age14	0.055***	(0.005)	-0.120***	[0.008]	0.063***	(0.005)	-0.075***	[0.005]
In-kind	0.029	(0.034)			0.052***	(0.016)		
Income restriction	-0.021**	(0.010)			-0.006	(0.015)		
R-squared	0.1	5			0.2	22		
Ν	998,	107			994,	982		
N of Individual	26,0	97			26,0)76		
Mean wo subsidy	0.2	20			0.4	4		
Mean w subsidy	0.3	2			0.5	56		

Table 8: Price responsiveness by health status

A. Outpatient visit dummy

Notes: An outpatient visit dummy takes one if an individual makes at least one outpatient visit per month and zero otherwise. We determine each patient's health status by the outpatient spending in the first 6 months after one is observed in the claim data. Then, we divide enrollees into two types (i.e., sicker or healthier) by the median spending for each age (in years) and the subsidy status. See the main text for the details. All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. For the estimates, the standard errors clustered at the municipality level are reported in parenthesis. For the arc-elasticities, the bootstrapped standard errors with 200 repetitions clustered at municipality level are reported in brackets. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Significance levels: *** p<0.01, ** p<0.05, * p<0.10

		Hea	althier			S	icker	
	(1))	(2)		(3)	(4)	
	Estimate	(SE)	Arc- elasticities	[SE]	Estimate	(SE)	Arc- elasticities	[SE]
Subsidized $ imes$								
Age7	1.085^{***}	(0.173)	-0.169***	[0.013]	2.007^{***}	(0.544)	-0.109***	[0.012]
Age8	1.128***	(0.132)	-0.179***	[0.012]	1.645***	(0.434)	-0.093***	[0.021]
Age9	1.160***	(0.127)	-0.194***	[0.014]	1.357***	(0.444)	-0.080***	[0.016]
Age10	1.228***	(0.145)	-0.213***	[0.015]	1.665***	(0.534)	-0.105***	[0.012]
Age11	1.219***	(0.153)	-0.218***	[0.015]	1.078**	(0.477)	-0.070***	[0.013]
Age12	1.130***	(0.128)	-0.202***	[0.016]	0.606	(0.458)	-0.039***	[0.014]
Age13	0.902^{***}	(0.123)	-0.159***	[0.015]	0.700^{*}	(0.423)	-0.044***	[0.014]
Age14	0.877^{***}	(0.109)	-0.159***	[0.015]	1.105***	(0.286)	-0.072***	[0.015]
In-kind	0.521	(0.589)			0.371	(0.656)		
Income restriction	-0.444	(0.381)			-0.629	(0.565)		
R-squared	0.2	28			0.5	52		
Ν	998,	107			994,	982		
N of Individual	26,0	97			26,0)76		
Mean wo subsidy	2.0)4			6.8	39		
Mean w subsidy	3.7	'0			9.8	34		

B. Outpatient spending per month (in 1K JPY)

Notes: The outpatient spending is total monthly spending on outpatient care measured in thousands JPY (roughly 10USD). We determine each patient's health status by the outpatient spending in the first 6 months after one is observed in the claim data. Then, we divide enrollees into two types (i.e., sicker or healthier) by the median spending for each age (in years) and the subsidy status. See the main text for the details. All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. For the estimates, the standard errors clustered at the municipality level are reported in parenthesis. For the arc-elasticities, the bootstrapped standard errors with 200 repetitions clustered at municipality level are reported in brackets. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Significance levels: *** p<0.01, ** p<0.05, * p<0.10

Online Appendix

(Not for Publication)

- Section A <u>Additional figures and tables</u>
- Section B <u>Elasticities</u>
- Section C <u>Event-study</u>
- Section D <u>Robustness Checks</u>
- Section E <u>Other outcomes</u>
- Section F By service categories
- Section G <u>Ambulatory Care Sensitive Conditions</u>
- Section H Child Mortality
- Section I By time of visits
- Section J Inappropriate use of antibiotics
- Section K Price responsiveness by health status
- Section L Inter-municipality migration

Appendix A: Additional figures and tables

Figure A-1: Timing of the subsidy expansions



A. By year-month

Notes: The total number of municipalities is 165. The data spans from April 2005 to March 2015 (10 years). There are total of 198 subsidy expansions for child health care.



Figure A-2: The number of observations by subsidy status

Notes: The two vertical dotted lines indicates the ages of children focused in this paper since we do not have many observations without subsidy below age 7 and with subsidy above age 15. This happens because majority of the municipalities (81.3%) have already provided subsidy till age 6 (the start of primary school) at the beginning of our sample period, and also most of the municipalities do not provide subsidy beyond age 15 (the end of junior high school) at the end of our sample period (April 2005–March 2015).

Figure A-3: Basic results (monthly)



A. Outpatient visit dummy

B. Outpatient spending (in 1K JPY)



Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). The estimates β_a (where *a* is age in months) are plotted. The dotted lines are the 95th confidence intervals and the standard errors clustered at municipality level are used to construct them. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization.

		Mun-time-	age cell	Year-month		
Before change	After change	Ν	%	Ν	%	
30%	0%	3,308	29.5%	14,166	38.4%	
0%	30%	2,768	24.7%	11,685	31.6%	
500 JPY/visit	30%	1,021	9.1%	2,489	6.7%	
30%	200 JPY/visit	786	7.0%	1,365	3.7%	
30%	500 JPY/visit	643	5.7%	1,425	3.9%	
200 JPY/visit	20%	471	4.2%	969	2.6%	
200 JPY/visit	0%	432	3.9%	865	2.3%	
200 JPY/visit	30%	331	3.0%	460	1.2%	
300 JPY/visit	30%	258	2.3%	479	1.3%	
10%	30%	243	2.2%	702	1.9%	
30%	300 JPY/visit	158	1.4%	308	0.8%	
10%	0%	149	1.3%	413	1.1%	
0%	10%	125	1.1%	422	1.1%	
300 JPY/visit	200 JPY/visit	118	1.1%	265	0.7%	
30%	10%	118	1.1%	247	0.7%	
0%	20%	51	0.5%	65	0.2%	
15%	0%	51	0.5%	217	0.6%	
15%	30%	37	0.3%	105	0.3%	
30%	20%	36	0.3%	46	0.1%	
30%	15%	33	0.3%	144	0.4%	
0%	200 JPY/visit	27	0.2%	30	0.1%	
0%	15%	17	0.2%	32	0.1%	
200 JPY/visit	300 JPY/visit	14	0.1%	14	0.0%	
500 JPY/visit	20%	8	0.1%	8	0.0%	
300 JPY/visit	0%	1	0.0%	1	0.0%	
20%	0%	1	0.0%	1	0.0%	
Total		11.205	100%	36.923	100%	

Table A-1: List of changes in subsidy status

Notes: This table lists all combinations of transitions in price cost-sharing in our original data. In this paper, we focus on the first two transitions. 200, 300 and 500 JPY are roughly USD2, 3, and 5, respectively.

	In our	Not in our	Dif
Variable			(2) - (1) (2)
variable	(1)	(2)	(3)=(1)-(2)
<u>Characteristics</u>			
Female	0.49	0.49	0.00
	[0.50]	[0.50]	(0.01)
Age (in years)	11.83	12.06	-0.23
	[2.86]	[2.71]	(0.15)
<u>Utilization</u>			
Outpatient visit dummy	0.32	0.32	0.01
	[0.47]	[0.46]	(0.01)
Outpatient spending (in 1K JPY)	4.47	4.20	0.27
	[21.56]	[16.40]	(0.20)
N of outpatient visits	0.62	0.57	0.05^{***}
	[1.26]	[1.17]	(0.01)
Inpatient admission dummy (×1000)	2.39	2.65	-0.26
	[48.81]	[51.42]	(0.26)
Inpatient spending (in 1K JPY)	0.98	1.28	-0.30
	[33.26]	[40.22]	(0.22)
N	660,697	301,005	
N of individuals	24,429	11,846	

Table A-2: Sample selection

Notes: The sample is limited to person-month observations *without* subsidy. Columns (1) and (2) report the means of variables in the far-left column for in our sample and not in our sample, respectively. The standard deviations are in brackets. Column (3) reports the difference in means between Columns (1) and (2) with standard errors clustered at the municipality in parentheses. The outpatient and inpatient spending are measured in thousands JPY (roughly 10USD). Significance levels: *** p < 0.01, ** p < 0.05, * p < 0.10

	ICD10	Share
A. Broad Category		
Diseases of the respiratory system	J00 – J99	31.4%
Diseases of the skin and subcutaneous tissue	L00 – L99	13.2%
Diseases of the eye and adnexa	H00 – H59	13.0%
Certain infectious and parasitic diseases	A00 – B99	10.0%
Diseases of the ear and mastoid process	H60 – H95	6.5%
Injury, poisoning and external causes	V01 - Y98	6.4%
<u>B. ICD10 4digit (Top 10)</u>		
Allergic rhinitis, unspecified	J304	9.5%
Acute bronchitis, unspecified	J209	4.9%
Asthma, unspecified	J459	4.8%
Acute atopic conjunctivitis	H101	4.1%
Acute sinusitis, unspecified	J019	3.6%
Acute laryngopharyngitis	J060	3.5%
Astigmatism	H522	3.1%
Dermatitis, unspecified	L309	2.5%
Acute pharyngitis, unspecified	J029	2.5%
Diarrhea and gastroenteritis of infectious origin	A09-	2.5%

Table A-3: List of diagnosis group and ICD10

Appendix B: Elasticities

B.1 Arc-elasticity

For our basic estimate that does not distinguish the asymmetry of price changes, we report the arc-elasticity. Our basic estimation equation [4] from the main text is repeated here as follows:

$$Y_{iamt} = \alpha + \sum_{A=7}^{14} \beta_A \{subsidized_{iamt} \times 1(Age A)\} + \gamma X'_{mt} + \delta_a + \pi_t + \theta_i + \varepsilon_{iamt} - [B1]$$

Then, the arc-elasticity for each age in year A is defined by

$$arc\text{-}elasticity_{A} = \left(\frac{Q_{1A} - Q_{0A}}{Q_{0A} + Q_{1A}}\right) / \left(\frac{P_{1A} - P_{0A}}{P_{1A} + P_{0A}}\right) = \left(\frac{\beta_{A}}{Q_{0A} + Q_{1A}}\right) / \left(\frac{0 - 0.3}{0.3 + 0}\right) = -\frac{\beta_{A}}{Q_{0A} + Q_{1A}} \quad -[B2]$$

where Q_{1A} , Q_{0A} are quantity of health care utilization for subsidized (denoted by 1), and unsubsidized (denoted by 0), and P_{1A} , P_{0A} are defined in the same way for prices. β_A are the estimates from the first equation. We report arcelasticity because they are widely used and comparable to estimates from RAND HIE, in which the largest plan was also the free care plan. However, one can make an argument that when the starting price is zero, as in our case, price elasticity may not be well defined. Thus, as an alternative, we also report the *semi* arc-elasticity, which is defined by

semi arc-elasticity_A =
$$\left(\frac{Q_{1A}-Q_{0A}}{Q_{0A}+Q_{1A}}\right) / \left(\frac{P_{1A}-P_{0A}}{2}\right) = \left(\frac{\beta_A}{Q_{0A}+Q_{1A}}\right) / \left(\frac{0-0.3}{2}\right) = arc-elasticity_A / 0.15 - [B3]$$

Thus, in our case, *semi* arc-elasticity is simply arc-elasticity divided by 0.15. For example, *semi* arc-elasticities are reported in Brot-Goldberg *et al.* (2017), and Nilsson and Paul (2015).

B.2 Semi point-elasticity

For the analysis of asymmetric responses, we instead report semi *point*-elasticity instead of *arc*-elasticity as we exactly know the starting quantity, and also the direction of the price changes. Again, the equation [5] from the main text is repeated here:

$$Y_{iamt} = \alpha + \sum_{A=7}^{14} \beta_A^{Better} \{subsidized_{iamt} \times better_{iamt} \times 1(Age A)\} + \sum_{A=7}^{14} \beta_A^{Worse} \{subsidized_{iamt} \times worse_{iamt} \times 1(Age A)\} + \gamma X'_{mt} + \delta_a + \pi_t + \theta_i + \varepsilon_{iamt} - [B4]$$

Then, the semi *point*-elasticity for each direction of price changes are defined as:

semi point-elasticity_A^{Better} =
$$\left(\frac{Q_{1A}-Q_{0A}}{Q_{0A}}\right)/(P_{1A}-P_{0A}) = -\left(\frac{\beta_A^{Better}}{Q_{0A}}\right)/0.3$$
 -[B4]
semi point-elasticity_A^{Worse} = $\left(\frac{Q_{0A}-Q_{1A}}{Q_{1A}}\right)/(P_{0A}-P_{1A}) = \left(\frac{\beta_A^{Worse}}{Q_{1A}}\right)/0.3$ -[B5]

where β_A^{Better} and β_A^{Worse} are estimates from equation [B4] above. For arc-elasticity and semi point-elasticity, the standard errors clustered at municipality are obtained by bootstrapping with 200 repetitions.

References:

- Brot-Goldberg, Zarek C., Amitabh Chandra, Benjamin R. Handel, and Jonathan T. Kolstad. (2017) "What does a Deductible Do? The Impact of Cost-Sharing on Health Care Prices, Quantities, and Spending Dynamics." *Quarterly Journal of Economics* 132(3): 1261–1318.
- Nilsson, Anton, and Alexander Paul. (2015) "The Effect of Copayments on Children's and Adolescents' Use of Medical Care." Unpublished manuscript.

Appendix C: Event-study

-24 -21 -18 -15 -12

-9 -6 -3 0 3 6 9

Months from changes



Figure C-1: Event study (±24 months)

Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). "Better" indicates the subsidy expansion which lowers the price of health care from 30% to 0%, and "Worse" indicates subsidy expiration that raises the price from 0% to 30%. The solid lines plot the estimates from a variant of estimation equation [4] where the subsidized dummy is replaced by a series of dummy for each month ranging from 12 months prior to the change in subsidy status to 24 months after the change in subsidy status (T= -24 to +23, where T = 0 is the change). The dotted lines are the 95th confidence interval derived from standard errors clustered at municipality level. The reference month is 3 months before the subsidy changes (T = -3). The observations within two months from the subsidy changes of the opposite direction are excluded from the sample to account for anticipatory utilization. Note that the scales of *y*-axis are set the same within the panels so that two figures for opposite directions of subsidy changes are visually comparable.

-24 -21 -18 -15 -12

-9

-6

-3

0 3 6

Months from changes

9 12

15 18 21

24

12 15 18

21 24

Figure C-2: Event study at selected ages





Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). "Better" indicates the subsidy expansion which lowers the price of health care from 30% to 0%, and "Worse" indicates subsidy expiration that raises the price from 0% to 30%. The solid lines plot the estimates from a variant of estimation equation [4] where the subsidized dummy is replaced by a series of dummy for each month ranging from 12 months prior to the change in subsidy status to 12 months after the change in subsidy status (T = -12 to +11, where T = 0 is the change) at any ages between 9 and 0 months and 9 and 11 months. The dotted lines are the 95th confidence interval derived from standard errors clustered at municipality level. The reference month is 3 months before the subsidy changes (T = -3). The observations within two months from the subsidy changes of the opposite direction are excluded from the sample to account for anticipatory utilization. Note that the scales of *y*-axis are set the same within the panels so that two figures for opposite directions of subsidy changes are visually comparable.

(ii) Age 12 A. Outpatient visit dummy



Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). "Better" indicates the subsidy expansion which lowers the price of health care from 30% to 0%, and "Worse" indicates subsidy expiration that raises the price from 0% to 30%. The solid lines plot the estimates from a variant of estimation equation [4] where the subsidized dummy is replaced by a series of dummy for each month ranging from 12 months prior to the change in subsidy status to 12 months after the change in subsidy status (T = -12 to +11, where T = 0 is the change) at any ages between 12 and 0 months and 12 and 11 months. The dotted lines are the 95th confidence interval derived from standard errors clustered at municipality level. The reference month is 3 months before the subsidy changes (T = -3). The observations within two months from the subsidy changes of the opposite direction are excluded from the sample to account for anticipatory utilization. Note that the scales of *y*-axis are set the same within the panels so that two figures for opposite directions of subsidy changes are visually comparable.

Appendix D: Robustness Checks

We subject the main results in Table 2 (and Figure 5) to a series of robustness checks. Critically, these results on the causal effects of patient cost-sharing to be robust across all specifications considered. Figure D-1 plots the estimates of key robustness checks together with our baseline estimates from Figure 5. The corresponding results are summarized in Tables C-1. For the ease the comparison, Columns (1) present the estimates from our baseline model of equation [4].

First, we address the potential concern that our control group—namely municipalities without changes in subsidy—exhibits a different time trend than municipalities with subsidy. For example, if the municipalities in better business cycles are more likely to implement the subsidy expansion, while income effects simply increase utilization, our estimates can be upward biased. Since the estimates in event-study before T=0 seems to be reasonably smooth and close to zero, this does not seem to be a serious concern. Nonetheless, we conduct several robustness checks to address this concern. Specifically, we add the municipality specific time trend in Column (2) and time-by-municipality FEs (where time is measured in months) in Column (3) to account for the time-varying municipality characteristics that are correlated with both the expansion of the subsidy and the health care utilization. The latter is especially stringent as these fixed effects capture any municipality specific policy change (e.g., income transfer or other subsidies) or event (outbreak of influenza) in a particular month. We are reassured that row (i) in Figure D-1 shows that the estimates are almost unchanged.

Another way to account for the potential endogeneity of subsidy expansion is to restrict the sample to only those children which experienced at least one change in subsidy status, thereby, dropping children which remain either subsidized or unsubsidized throughout the sample period. Importantly, this identification strategy only exploits the *timing* of the changes in subsidy status. In this way, we can to some extent mitigate the concern that individuals in the treatment and control groups are different. Row (ii) in Figure D-1 shows that the estimates are somewhat noisier due to smaller sample but are qualitatively similar.

Third, we collapse the data at municipality-age-time cells, which is the level of variation, to partially account for zero spending at the person-month level. Then, we estimate analogous to equation [4] where the number of observations in each cell is used as a weight. Row (iii) in Figure D-1 presents the estimates. It is reassuring that the estimates from the cell level analysis yields almost identical results to those from underlying individual micro data.

As a separate exercise, Figure D-2 and Table D-2 presents the sensitivity of our estimates to the size of the "donut-hole". The estimates and hence elasticities are barely affected after excluding 2 months from both sides of T=0.



Figure D-1: Robustness checks (Estimates only)

Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). The dotted lines are the 95th confidence intervals derived from standard errors clustered at municipality level except for 2) where standard errors clustered at individual level. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Along with our baseline estimates, (i) reports the estimates which include time by municipality FE, (ii) reports the estimates from the sample limited to those individuals which experienced at least one change in subsidy status, and (iii) reports the estimates from cell means where the cell is defined by municipality-age-time and the number of observations in each cell is used as a weight.



Figure D-2: Sizes of "donut" holes and the estimates/elasticities

Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). The lines plot the estimates from the sample where the observations within one, two, and three months from subsidy changes are excluded along with the estimates of no exclusion ("no hole").

Table D-1: Robustness checks

(1) Outcome. Outputient visit duminy	(i)) Outcome:	Outpatient	visit	dummy
--------------------------------------	-----	------------	------------	-------	-------

	Baseline (Col. 1 Table 2)		With municipality- time trend		With municipality- time FE		Among on	ly treated	Cell		
	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	
	(1)	(2)		(3)	(3)		(4)		(5)	
Subsidized $ imes$											
Age7	0.075^{***}	(0.006)	0.070^{***}	(0.006)	0.071^{***}	(0.006)	0.070^{***}	(0.008)	0.073***	(0.006)	
Age8	0.068^{***}	(0.006)	0.064^{***}	(0.005)	0.063***	(0.005)	0.070^{***}	(0.009)	0.068^{***}	(0.006)	
Age9	0.070^{***}	(0.005)	0.066^{***}	(0.005)	0.065***	(0.005)	0.063***	(0.007)	0.070^{***}	(0.005)	
Age10	0.073***	(0.005)	0.070^{***}	(0.005)	0.068^{***}	(0.006)	0.074^{***}	(0.007)	0.074^{***}	(0.005)	
Age11	0.066^{***}	(0.006)	0.063***	(0.006)	0.061***	(0.006)	0.071***	(0.007)	0.067^{***}	(0.005)	
Age12	0.064^{***}	(0.004)	0.062***	(0.005)	0.060^{***}	(0.006)	0.069***	(0.006)	0.065***	(0.004)	
Age13	0.062***	(0.004)	0.061***	(0.004)	0.058^{***}	(0.005)	0.065***	(0.005)	0.061***	(0.005)	
Age14	0.060^{***}	(0.004)	0.060^{***}	(0.005)	0.057^{***}	(0.005)	0.069***	(0.009)	0.059***	(0.005)	
In-kind	0.047^{***}	(0.014)	0.054^{***}	(0.019)	0.017	(0.031)	0.040^{**}	(0.019)	0.054^{***}	(0.016)	
Income restriction	-0.020**	(0.009)	-0.015*	(0.008)	-0.014	(0.009)	-0.017^{*}	(0.009)	-0.017**	(0.008)	
R-squared	0.2	23	0.23		0.24		0.19		0.30		
N	2,205	,647	2,205	,647	2,204	,496	862,211		465,241		
N of Individual	63,530		63,5	30	63,5	63,502		92	-		
Age FE	Х	X X			Х	-	Х	-	Х		
Time FE	Х		Х	Х		Х		Х		Х	
Individual FE	Х		Х	X X			Х		-		
Municipality-Time tren	nd		Х	•							
Municipality-Time FE					Х						

Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month. For ease of comparison, Column (1) replicates the estimates from Column (1) in Table 2. Columns (2) and (3) add municipality-specific trend and municipality-time (in months) FE, respectively. Column (4) reports the estimates from the sample limited to those individuals which experienced at least one change in subsidy status. Column (5) reports the estimates from cell means where the cell is defined by municipality-age-time and the number of observations in each cell is used as a weight. The standard errors clustered at the municipality level are reported in parenthesis. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. *** p<0.01, ** p<0.05, * p<0.10

(ii) Outcome: Outpatient spending (in 1K JPY)

	Baseline (Col. 3 Table 2)		With municipality- time trend		With mun time	With municipality- time FE		ly treated	Cell		
	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	
	(1)	(2)		(3	(3)		(4)		(5)	
Subsidized X											
Age7	1.376***	(0.236)	1.452***	(0.216)	1.422***	(0.253)	1.401^{***}	(0.285)	1.267***	(0.204)	
Age8	1.278^{***}	(0.179)	1.366***	(0.172)	1.331***	(0.201)	1.293***	(0.333)	1.310***	(0.213)	
Age9	1.131***	(0.171)	1.215***	(0.147)	1.180^{***}	(0.186)	1.073***	(0.297)	1.269***	(0.216)	
Age10	1.350***	(0.221)	1.430***	(0.189)	1.378***	(0.249)	1.353***	(0.315)	1.468***	(0.251)	
Age11	1.113***	(0.211)	1.184***	(0.191)	1.128***	(0.237)	1.196***	(0.287)	1.157***	(0.231)	
Age12	0.872***	(0.223)	0.943***	(0.212)	0.944^{***}	(0.257)	0.741^{**}	(0.314)	0.900^{***}	(0.208)	
Age13	0.811***	(0.218)	0.889^{***}	(0.196)	0.912***	(0.260)	0.602	(0.435)	0.907^{***}	(0.186)	
Age14	0.998***	(0.134)	1.069***	(0.124)	1.120***	(0.197)	1.531***	(0.414)	1.006***	(0.167)	
In-kind	0.440	(0.388)	0.815***	(0.305)	0.405	(0.792)	0.097	(0.336)	0.399	(0.363)	
Income restriction	-0.561	(0.372)	-0.519*	(0.269)	-0.366	(0.288)	-0.501	(0.439)	-0.297	(0.617)	
R-squared	0.5	51	0.51		0.51		0.54		0.56		
Ν	2,205,647		2,205	,647	2,204	,496	862,2	862,211		465,241	
N of Individual	63,530		63,5	30	63,5	02	13,8	92	-		
Age FE	Х	Х		-	Х	-	Х	-	Х		
Time FE	Х		Х	Х		Х		Х			
Individual FE	Х		Х		Х	Х		-	-		
Municipality-Time tren	nd		Х	-							
Municipality-Time FE					Х	-					

Notes: The outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). For ease of comparison, Column (1) replicates the estimates from Column (1) in Table 2. Columns (2) and (3) add municipality-specific trend and municipality-time (in months) FE, respectively. Column (4) reports the estimates from the sample limited to those individuals which experienced at least one change in subsidy status. Column (5) reports the estimates from cell means where the cell is defined by municipality-age-time and the number of observations in each cell is used as a weight. The standard errors clustered at the municipality level are reported in parenthesis. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. *** p<0.01, ** p<0.05, * p<0.10

	No exclusion		± 1 month ± 2 months excluded excluded		±3months excluded		No exclusion	±1month excluded	±2months excluded	±3months excluded		
	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)		Arc-eld	asticities	
	(1))	(2)	(3)	(4	-)	(5)	(6)	(7)	(8)
Subsidized $ imes$												
Age7	0.084^{***}	(0.005)	0.077^{***}	(0.006)	0.075^{***}	(0.006)	0.076^{***}	(0.007)	-0.106	-0.097	-0.095	-0.095
Age8	0.077^{***}	(0.004)	0.069***	(0.005)	0.068^{***}	(0.006)	0.067^{***}	(0.007)	-0.098	-0.088	-0.086	-0.086
Age9	0.080^{***}	(0.004)	0.072***	(0.005)	0.070^{***}	(0.005)	0.069***	(0.006)	-0.105	-0.093	-0.091	-0.090
Age10	0.083***	(0.005)	0.074^{***}	(0.005)	0.073***	(0.005)	0.073***	(0.005)	-0.109	-0.097	-0.095	-0.095
Age11	0.076^{***}	(0.005)	0.067^{***}	(0.006)	0.066***	(0.006)	0.065***	(0.006)	-0.101	-0.089	-0.086	-0.085
Age12	0.076^{***}	(0.004)	0.067^{***}	(0.004)	0.064^{***}	(0.004)	0.062***	(0.005)	-0.100	-0.087	-0.083	-0.081
Age13	0.071***	(0.004)	0.064^{***}	(0.004)	0.062***	(0.004)	0.060^{***}	(0.004)	-0.092	-0.082	-0.078	-0.076
Age14	0.071***	(0.004)	0.063***	(0.004)	0.060^{***}	(0.004)	0.059***	(0.005)	-0.090	-0.079	-0.074	-0.073
In-kind	0.041***	(0.013)	0.042***	(0.013)	0.047^{***}	(0.014)	0.051***	(0.014)				
Income restriction	-0.021***	(0.008)	-0.018**	(0.008)	-0.020**	(0.009)	-0.017	(0.011)				
R-squared	0.2	3	0.2	23	0.2	23	0.2	23				
N	2,303	,335	2,253	,851	2,205	,647	2,158	3,881				
N of Individual	63,5	90	63,5	570	63,5	530	63,3	362				

Table D-2: The size of "donut" holes and corresponding estimates/elasticity

(i) Outcome: Outpatient visit dummy

Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month. All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. The standard errors clustered at the municipality level are reported in parenthesis. Column (1) does not exclude any observations, while Columns (2)–(4) exclude the observations within one, two, and three months from the both sides of subsidy changes to account for anticipatory utilization. Columns (5)–(8) report the arc-elasticities that corresponds to estimates from Columns (1)–(4). Significance levels: *** p<0.01, ** p<0.05, * p<0.10

(ii) Outcome: Outpatient spending (in 1K JPY)

	No exclusion	± 1 month ± 2 months excluded excluded		±3months excluded	No ± exclusion e	±1month excluded	±2months excluded	±3months excluded
	Estimate (SE)	Estimate (SE)	Estimate (SE)	Estimate (SE)		Arc-elas	sticities	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Subsidized $ imes$								
Age7	1.812*** (0.192)	1.493*** (0.221)	1.376*** (0.236)	1.357*** (0.270)	-0.158	-0.131	-0.121	-0.117
Age8	1.705*** (0.164)	1.410*** (0.175)	1.278*** (0.179)	1.249*** (0.199)	-0.153	-0.126	-0.115	-0.110
Age9	1.565*** (0.155)	1.243*** (0.163)	1.131*** (0.171)	1.083*** (0.196)	-0.142	-0.112	-0.102	-0.097
Age10	1.749*** (0.200)	1.453*** (0.217)	1.350*** (0.221)	1.319*** (0.246)	-0.160	-0.131	-0.121	-0.118
Age11	1.502*** (0.192)	1.214*** (0.208)	1.113*** (0.211)	1.062*** (0.229)	-0.137	-0.108	-0.099	-0.094
Age12	1.255**** (0.186)	0.970**** (0.201)	0.872*** (0.223)	0.824*** (0.253)	-0.109	-0.083	-0.074	-0.070
Age13	1.117**** (0.192)	0.881*** (0.205)	0.811*** (0.218)	0.748*** (0.240)	-0.094	-0.072	-0.066	-0.061
Age14	1.341*** (0.112)	1.080**** (0.117)	0.998*** (0.134)	0.899*** (0.166)	-0.113	-0.088	-0.080	-0.073
In-kind	-0.385 (0.661)	0.422 (0.354)	0.440 (0.388)	0.470 (0.435)				
Income restriction	-0.564* (0.317)	-0.553 (0.356)	-0.561 (0.372)	-0.626 (0.401)				
R-squared	0.51	0.51	0.51	0.51				
N	2,303,335	2,253,851	2,205,647	2,158,881				
N of Individual	63,590	63,570	63,530	63,362				

Notes: The outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. The standard errors clustered at the municipality level are reported in parenthesis. Column (1) does not exclude any observations, while Columns (2)–(4) exclude the observations within one, two, and three months from subsidy changes to account for anticipatory utilization. Columns (5)–(8) report the arc-elasticities that corresponds to estimates from Columns (1)–(4). Significance levels: *** p<0.01, ** p<0.05, * p<0.10

Appendix E: Other outcomes



Figure E-1: Other outcomes

Notes: The frequency of outpatient visits is the number of outpatient visits per month. The outpatient spending per visit measured in thousand JPY (roughly USD10) is the monthly outpatient spending divided by the number of outpatient visits per month. The dotted lines are the 95th confidence intervals. The standard errors clustered at municipality level are used for estimates, and the bootstrapped standard errors clustered at municipality with 200 repetitions are used for the arc-elasticities. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization.
	A. Fr	f outpatient vi	B. O	B. Outpatient spending <i>per</i> visit (in 1K JPY)				
	(1))	(2)		(3))	(4)	
	Estimate	(SE)	Arc- elasticities	[SE]	Estimate	(SE)	Arc- elasticities	[SE]
Subsidized $ imes$								
Age7	0.217***	(0.026)	-0.110***	[0.006]	0.467^{***}	(0.110)	-0.076***	[0.010]
Age8	0.213***	(0.019)	-0.118***	[0.006]	0.393***	(0.107)	-0.063***	[0.012]
Age9	0.196***	(0.016)	-0.117***	[0.006]	0.332**	(0.137)	-0.054***	[0.011]
Age10	0.200^{***}	(0.016)	-0.128***	[0.006]	0.488^{***}	(0.158)	-0.082***	[0.010]
Age11	0.181***	(0.020)	-0.126***	[0.006]	0.438***	(0.119)	-0.075***	[0.011]
Age12	0.157***	(0.016)	-0.118***	[0.006]	0.368***	(0.127)	-0.06***	[0.013]
Age13	0.152***	(0.016)	-0.120***	[0.006]	0.279^*	(0.144)	-0.044***	[0.013]
Age14	0.140^{***}	(0.017)	-0.116***	[0.007]	0.431***	(0.137)	-0.069***	[0.014]
In-kind	0.072	(0.051)			0.141	(0.190)		
Income restriction	-0.083***	(0.026)			-0.141	(0.208)		
R-squared	0.2	8			0.4	4		
Ν	2,205,	647			2,205	,647		
N of Individual	63,5	30			63,5	30		
Mean wo subsidy	0.6	2			2.5	9		

Table E-1: Other outcomes

Notes: The frequency of outpatient visits is the number of outpatient visits per month. The outpatient spending per visit measured in thousand JPY (roughly USD10) is the monthly outpatient spending divided by the number of outpatient visits per month. All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. For the estimates, the standard errors clustered at the municipality level are reported in parenthesis. For the arc-elasticities, the bootstrapped standard errors with 200 repetitions clustered at municipality level are reported in brackets. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Significance levels: *** p<0.01, ** p<0.05, * p<0.10

Appendix F: By service categories

Given we find the substantial increases in outpatient utilization, it is natural to ask which types of medical services are driving the results. To do so, we group the medical services into six broad categories: medication, consultation fees, laboratory tests, non-surgical procedure, surgical procedure, and others. The last row in Table F-1 shows that the medication is by far the largest share in outpatient spending that accounts more than half (54.1%). Note here that medication includes fees not only for medicine itself but also related to prescribing and dispensing medications, including fees at the pharmacy. Consultation fees (18.4%), laboratory tests (17.2%), and non-surgical procedure (5.3%) are next three categories, and combined with medication, these four categories account for 95% of total outpatient spending. The remaining categories are "others" (3.4%) and "surgical procedure" (1.6%). Thus, in what follows, we focus on these four main service categories.

Figure F-1 plots the event study separately for these four service categories on outpatient spending. As expected, we see the abrupt changes in utilization at the time of subsidy changes for all service categories examined. In addition, as consistent with intuition, the magnitudes of anticipatory spending seem to large in mediation and laboratory tests, suggesting that these medical services can be more easily timed (stockpiled) than other services such as non-surgical procedure.

Figure F-2 provides the graphical presentation of our difference-in-difference estimates from equation [4]. Table F-1 is the corresponding table. The consultation fees—which are charged in each visit and thus is closely related to the frequency—are least price sensitive. On the other hand, the medical services related to the treatment intensity such as laboratory tests and non-surgical procedures are more price sensitive.¹ This result is consistent with our finding that not only the frequency of visits but also the spending per visit increases. Interestingly, the medication is not as price sensitive as other service categories. We also find that brand-name drugs and generics are similarly price sensitive (results available upon request).

References:

Lee, David, and Frank Levy. (2012) "The Sharp Slowdown in Growth of Medical Imaging: An Early Analysis Suggests Combination of Policies Was the Cause." *Health Affairs* 31(8): 1–9.

¹ Laboratory test include imaging, which is often identified as having unproven medical value (e.g., Lee and Levy 2012). When imaging is separately examined, we also find the statistically significant increase as well (results available upon request).

Figure F-1: Event study by service categories (Outpatient spending (in 1K JPY)



Notes: The outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). "Better" indicates the subsidy expansion which lowers the price of health care from 30% to 0%. The solid lines plot the estimates from a variant of estimation equation [4] where the subsidized dummy is replaced by a series of dummy for each month ranging from 12 months prior to the change in subsidy status to 12 months after the change in subsidy status (T= -12 to +11, where T= 0 is the change) at age between 12 and 0 months and 12 and 11 months. The dotted lines are the 95th confidence interval derived from standard errors clustered at municipality level. The reference month is 3 months before the subsidy changes (T= -3). The observations within two months from the subsidy changes of the opposite direction are excluded from the sample to account for anticipatory utilization.

(ii) Worse (subsidy expiration)



Notes: The outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). "Worse" indicates subsidy expiration that raises the price from 0% to 30%. The solid lines plot the estimates from a variant of estimation equation [4] where the subsidized dummy is replaced by a series of dummy for each month ranging from 12 months prior to the change in subsidy status to 12 months after the change in subsidy status (T= -12 to +11, where T= 0 is the change) at age between 12 and 0 months and 12 and 11 months. The dotted lines are the 95th confidence interval derived from standard errors clustered at municipality level. The reference month is 3 months before the subsidy changes (T= -3). The observations within two months from the subsidy changes of the opposite direction are excluded from the sample to account for anticipatory utilization.

Figure F-2: By service categories

Outcome: Outpatient spending (in 1K JPY)



Notes: Outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). The dotted lines are the 95th confidence intervals. The standard errors clustered at municipality level are used for estimates, and the bootstrapped standard errors clustered at municipality with 200 repetitions are used for the arc-elasticities. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Note that the scales of *y*-axis on the bottom half are set the same so that elasticities across service categories are visually comparable.

	A. Medication		B. Consult	ation Fees	C. Labora	tory tests	D. Non-surgical procedure		
	Estimate	Arc - elasticity	Estimate	Arc - elasticity	Estimate	Arc - elasticity	Estimate	Arc - elasticity	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Subsidized $ imes$									
Age7	0.518 ^{***} (0.182)	-0.096 ^{***} [0.015]	0.265^{***} (0.051)	-0.094 ^{***} [0.006]	0.274^{***} (0.040)	-0.148 ^{***} [0.011]	0.141^{***} (0.023)	-0.178 ^{***} [0.011]	
Age8	0.496 ^{***} (0.157)	-0.092 ^{***} [0.017]	0.190^{***} (0.054)	-0.071 ^{***} [0.006]	0.245 ^{***} (0.039)	-0.136 ^{***} [0.011]	0.133 ^{***} (0.018)	-0.183 ^{***} [0.012]	
Age9	0.392 ^{**} (0.160)	-0.075 ^{***} [0.018]	0.123 ^{**} (0.055)	-0.050 ^{***} [0.006]	0.277 ^{***} (0.036)	-0.153 ^{***} [0.011]	0.116 ^{***} (0.019)	-0.172 ^{***} [0.014]	
Age10	0.582 ^{***} (0.201)	-0.119 ^{***} [0.016]	0.148 ^{**} (0.059)	-0.063*** [0.006]	0.272 ^{***} (0.029)	-0.151*** [0.011]	0.096 ^{***} (0.017)	-0.157 ^{***} [0.016]	
Age11	0.392 ^{**} (0.190)	-0.081 ^{***} [0.018]	0.142 ^{**} (0.055)	-0.065*** [0.006]	0.219 ^{***} (0.028)	-0.125 ^{***} [0.012]	0.083 ^{***} (0.022)	-0.144 ^{***} [0.016]	
Age12	0.258 (0.221)	-0.050 ^{***} [0.016]	0.121 ^{***} (0.042)	-0.058 ^{***} [0.006]	0.194 ^{***} (0.027)	-0.107 ^{***} [0.011]	0.068 ^{***} (0.016)	-0.132 ^{***} [0.016]	
Age13	0.184 (0.211)	-0.034* [0.018]	0.168^{***} (0.036)	-0.083 ^{***} [0.006]	0.213 ^{***} (0.031)	-0.115*** [0.01]	0.056^{***} (0.015)	-0.121 ^{***} [0.017]	
Age14	0.434 ^{***} (0.137)	-0.082 ^{***} [0.022]	0.192 ^{***} (0.037)	-0.100 ^{***} [0.006]	0.201 ^{***} (0.036)	-0.112*** [0.011]	0.056 ^{***} (0.012)	-0.133 ^{***} [0.017]	
In-kind	0.239 (0.279)		0.152 ^{**} (0.076)		0.014 (0.071)		0.068 (0.043)		
Income restriction	-0.165 (0.269)		-0.030 (0.100)		-0.123*** (0.044)		-0.103 ^{***} (0.025)		
R-squared	0.53		0.25		0.12		0.17		
N	2,205,647		2,205,647		2,205,647		2,205,647		
N of Individual	63,530		63,530		63,530		63,530		
Mean wo subsidy	2.13		0.72		0.68		0.21		
Share	54.1%		18.4%		17.2%		5.3%		

Table F-1: By service categories

Outcome: Outpatient spending per month (in 1K JPY)

Notes: The outpatient spending is total monthly spending on outpatient care measured in thousands JPY (roughly 10USD). All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. For the estimates, the standard errors clustered at the municipality level are reported in parenthesis. For the arc-elasticities, the bootstrapped standard errors with 200 repetitions clustered at municipality level are reported in brackets. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. The remaining service categories are "Other" (3.4%), and "Surgery" (1.6%). Significance levels: *** p<0.01, ** p<0.05, * p<0.10

Name of diseases	A dummy	A dummy for disease		ICD-10
	Uncond.	Cond. on	among	
		visit	ACSC	
	Mean	Mean	(%)	
Congenital syphilis	0.000	0.000	0.0%	A50.0-A50.9
Immunization preventable conditions	0.002	0.005	1.0%	A35, A36, A37, A80, G00
Grand mal status and other epileptic convulsions	0.005	0.012	2.4%	G40, G41
Convulsions "A" & "B"	0.001	0.004	0.7%	R56
Severe ENT infections	0.114	0.280	56.9%	H66, H67, J02, J03, J06, J31.2
Bacterial pneumonia	0.003	0.007	1.5%	J13, J14, J15.3, J15.4, J15.7, J15.9, J16.8, J18, J18.1
Asthma	0.063	0.155	31.5%	J45, J46
Tuberculosis	0.000	0.000	0.0%	A15, A16, A17, A18, A19
Cellulitis	0.005	0.013	2.6%	L03 L04 L08.0 L08.8 L08.9 L88 L98.0
Diabetes "A", "B", "C"	0.000	0.001	0.1%	E10.0-E10.8, E11.0-E11.8, E12.0- E12.8, E13.0-E13.8, E14.0-E14.8
Hypoglycemia	0.000	0.001	0.2%	E16.2
Gastroenteritis	0.001	0.001	0.3%	K52.2, K52.8, K52.9
Kidney/urinary infection	0.000	0.001	0.2%	N10, N11, N12, N13.6
Dehydration-volume depletion	0.003	0.008	1.5%	E86
Iron deficiency anemia	0.002	0.005	0.9%	D50.1, D50.8, D50.9
Nutritional deficiencies	0.000	0.001	0.2%	E40, E41, E42, E43, E55.0, E64.3
Failure to thrive	0.000	0.000	0.0%	R629
Any ACSC	0.166	0.407		

Appendix G: Ambulatory Care Sensitive Conditions (ACSC)

Table G-1: List of ACSC

Notes: "Unconditional" includes observations (person-month) with no outpatient visits, and "Conditional" limits to observations with outpatient visits. We convert the ICD9-CM listed in Gadomski *et al.* (1998) to ICD10. *Source:* Gadomski *et al.* (1998)

Figure G-1: Arc-elasticities for Ambulatory Care Sensitive Conditions (ACSC)



Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and an inpatient admission dummy takes one if there is at least one hospitalization per month ($\times 1000$). See Table G-1 for the list of ACSC and summary statistics. The dotted lines are the 95th confidence intervals and bootstrapped standard errors clustered at municipality with 200 repetitions are used to construct them. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. ENT stands for Ear, Nose, and Throat. See Figure 7 in the main text for the corresponding estimates from equation [4].

Appendix H: Child Mortality

Figure H-1: Child Mortality



Notes: The estimates come from complementary log-log regression model where the baseline hazard is the log in age in months. The dotted lines are the 95th confidence intervals derived from standard errors clustered at municipality level. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization.

Baseline hazard	Log in age in months		Linec age in t	ar in months	Each dummy for age in years		
	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	
	(1)	(2	2)	(3)	
Subsidized X							
Age7	-1.336	(1.149)	-1.312	(1.132)	-1.433	(1.129)	
Age8	0.576	(0.559)	0.613	(0.570)	0.195	(0.628)	
Age9	-0.317	(0.510)	-0.275	(0.539)	-0.150	(0.541)	
Age10	-0.757	(0.795)	-0.716	(0.799)	0.210	(0.923)	
Age11	0.591*	(0.356)	0.624^{*}	(0.364)	0.694	(0.506)	
Age12	-1.469	(0.984)	-1.448	(0.984)	-1.582*	(0.916)	
Age13	0.348	(0.502)	0.353	(0.504)	0.420	(0.675)	
Age14	0.342	(0.500)	0.327	(0.498)	0.568	(0.566)	
N	2,205	,647	2,205	5,647	2,205	,647	
N of Individual	63,5	30	63,5	530	63,530		
N of deaths	68		68	8	68		

Table H-1: Child Mortality

Notes: The estimates come from complementary log-log regression models with different baseline hazard indicated at the first rows. The standard errors clustered at the municipality level are reported in parenthesis. *** p < 0.01, ** p < 0.05, * p < 0.10

Appendix I: By time of visits

Billing code #	Type of visit	Timing of visit	Additional fee charged (in 1K JPY)	Number of times charged (frequency)
Off-hour visits				
111000570	first visit	off-hour	0.85	4,329
111000870	first visit	off-hour ([*])	2.30	5,527
111012470	first visit	night/early morning	0.50	53,083
111700870	first visit	off-hour ([*])	2.30	5
112001110	revisit	off-hour	0.65	2,411
112001410	revisit	off-hour (*)	1.80	70
112006470	revisit	off-hour	0.65	256
112006770	revisit	off-hour (*)	1.80	1,986
112015570	revisit	night/early morning	0.50	62,423
<u>Midnight/Holiday v</u>	<u>visits</u>			
111000670	first visit	holiday	2.50	16,406
111000770	first visit	midnight	4.80	2,343
112001210	revisit	holiday	1.90	930
112001310	revisit	midnight	4.20	59
112006570	revisit	holiday	1.90	1,143
112006670	revisit	midnight	4.20	506

 Table I-1: List of billing codes for off-hour and midnight/holiday visits and corresponding additional fees charged

Notes: The fees listed on the far right are the fees charged additionally to the fees for regular-hour visits. As a benchmark, the fees for regular-hour visits during the sample period are roughly 2.8 for first visit, and 0.7 for revisits measured in thousands JPY (roughly USD10). (*) is applied to specific medical institutions. The unit of additional fee is in thousand JPY. *Source:* From Japan Federation of Democratic Medical Institutions (2015)

Figure I-1: Age profiles of frequency of visits by time of visits and subsidy status



Notes: Panel A plots the monthly frequency of visits for regular-hour visits, and Panel B plots the monthly frequency of visits for offhour and midnight/holiday visits. The dotted lines are age profiles of utilization without subsidy (labeled "no subsidy"), and the solid lines are age profiles of utilization with subsidy (labeled "subsidized").

Figure I-2: By time of visits

Outcome: Outpatient spending (in 1K JPY)

A. Regular-hour visits

B. Off-hour visits





Notes: The outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). See Table H-1 which provides the list of billing codes for these urgent visits and corresponding fees that are additionally charged on top of fees for regular-hour visits. The dotted lines are the 95th confidence intervals. The standard errors clustered at municipality level are used for estimates, and the bootstrapped standard errors clustered at municipality with 200 repetitions are used for the arc-elasticities. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization.

Table I-2: By time of visits

-	A. Regular-hour visits			-	B. Off-hour visits				C. Midnight/Holiday visits			
	(1)	(2))	(3)	(4))	(5)	(6)
	Estimate	(SE)	Arc- elasticity	[SE]	Estimate	(SE)	Arc- elasticity	[SE]	Estimate	(SE)	Arc- elasticity	[S E]
Subsidized $ imes$												
Age7	0.182***	(0.048)	-0.094***	[0.006]	0.015	(0.011)	-0.063***	[0.018]	0.007	(0.005)	-0.053	[0.033]
Age8	0.140***	(0.049)	-0.077***	[0.006]	0.006	(0.011)	-0.025	[0.017]	0.004	(0.004)	-0.031	[0.034]
Age9	0.093*	(0.049)	-0.055***	[0.006]	0.011	(0.010)	-0.048**	[0.019]	-0.004	(0.004)	0.036	[0.038]
Age10	0.101**	(0.048)	-0.062***	[0.007]	0.019**	(0.009)	-0.079***	[0.018]	-0.000	(0.005)	0.003	[0.039]
Age11	0.088^{**}	(0.043)	-0.059***	[0.007]	0.026***	(0.009)	-0.112***	[0.018]	-0.001	(0.004)	0.009	[0.039]
Age12	0.079**	(0.034)	-0.056***	[0.007]	0.026***	(0.006)	-0.108***	[0.016]	-0.005	(0.005)	0.059	[0.037]
Age13	0.097^{***}	(0.027)	-0.071***	[0.007]	0.043***	(0.008)	-0.159***	[0.016]	-0.001	(0.004)	0.011	[0.038]
Age14	0.118***	(0.025)	-0.092***	[0.007]	0.036***	(0.008)	-0.144***	[0.016]	0.003	(0.004)	-0.033	[0.040]
In-kind	0.080	(0.057)			0.052	(0.032)			-0.004	(0.019)		
Income restriction	-0.029	(0.055)			-0.008	(0.020)			-0.002	(0.009)		
R-squared	0.2	21			0.1	0			0.0)4		
Ν	2,205	,647			2,205	,647			2,205	,647		
N of Individual	63,5	530			63,5	530			63,5	530		
Mean wo subsidy	0.5	09			0.0	57			0.0	32		
Share	85.1	1%			9.5	%			5.4	%		

Notes: The outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). All the regressions include age (in months) FE, time (in month) FE, and individual FE. In-kind is a dummy that takes one if the municipality offers the subsidy in the form of in-kind instead of refund, and income restriction is a dummy that takes one if the municipality imposes income restriction for subsidy eligibility. For the estimates, the standard errors clustered at the municipality level are reported in parenthesis. For the semi point-elasticity, the bootstrapped standard errors with 200 repetitions clustered at municipality level are reported in brackets. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Significance levels: *** p<0.01, ** p<0.05, * p<0.10

Appendix J: Inappropriate use of antibiotics

Fleming-Dutra *et al.* (2016) divide the diagnoses (with corresponding ICD9-CM) into three tiers by the degree of appropriateness of antibiotics use. Specifically, Tiers 1, 2, and 3 are diagnostic categories where antibiotic use is always indicated, is occasionally indicated, and not indicated at all, respectively. For example, antibiotic prescription is considered appropriate for pneumonia because the diagnosis almost always warrants antibiotic therapy (Tier 1), while antibiotics for bronchitis and asthma are considered inappropriate because children with these conditions should not receive antibiotics (Tier 3).

In this paper, we focus on Tier 3 children who should not receive the antibiotics. When a patient has multiple diagnoses in a month, a priority is given to Tier 1 diagnoses, then Tier 2 diagnoses, then finally Tier 3 diagnoses so that a patient at each month is assigned to mutually exclusive tiers. Specifically, we assign a patient to Tier 1 when the patient has any diagnosis in Tier 1 in the month and to Tier 2 when the patient has any diagnosis in Tier 2 but not Tier 1, and the rest to Tier 3. In this way, Tier 3 only includes the patients for whom antibiotics should not be recommended at all since none of the diagnoses include the ones from Tier 1 and Tier 2. Table I-1 presents the list of Tier 3 diagnoses with corresponding ICD10 (which we convert from ICD9-CM) as well as summary statistics of antibiotic usage. The summary statistics for Tier and Tier 2 children are available upon request.

				Unconditiona	.1	Conditional on having the diagnosis			
Name of diagnosis	ICD 10	Fraction of the diagnosis (1)	Antibiotics use (dummy) (2)	Spending on antibiotics (in 1K JPY) (3)	Freq. of antibiotics prescriptions (4)	Antibiotics use (dummy) (5) = (2)/(1)	Spending on antibiotics (in 1K JPY) (6) = (3)/(1)	Freq. of antibiotics prescriptions (7) = $(4)/(1)$	
All		0.218	0.041	0.053	0.205	0.19	0.24	0.94	
By diagnosis:									
Asthma, allergy	J30, J44, J45, T784	0.058	0.015	0.021	0.084	0.27	0.36	1.45	
Bronchitis, bronchiolitis	J20, J21, J40	0.035	0.023	0.032	0.118	0.66	0.90	3.38	
Influenza	J09, J10, J11	0.019	0.007	0.009	0.034	0.37	0.48	1.80	
Non-suppurative otitis media	Н65, Н68, Н69	0.002	0.001	0.001	0.004	0.27	0.38	1.73	
Viral pneumonia	J12	0.000	0.000	0.000	0.000	0.33	0.50	2.34	
Viral upper respiratory infection	J00, J04, J05, J06, R05	0.033	0.017	0.022	0.087	0.53	0.67	2.65	
Other respiratory conditions	All remaining respiratory conditions (J00-J99) not coded above and R060-R064, R068- R069, R042, R048, R049, R093	0.000	0.000	0.000	0.000	0.26	0.30	1.78	
All other codes not listed elsewhere	All other codes not listed elsewhere	0.168	0.019	0.024	0.095	0.11	0.14	0.56	

Table J-1: Summary statistics of inappropriate antibiotic use

Notes: The spending on antibiotics is measured in thousand JPY (roughly USD10). The list of ICD10 codes comes from Fleming-Dutra *et al.* (2016) eTable "2. Diagnostic categories by tier with corresponding ICD-9CM code".

Appendix K: Price responsiveness by health status

In the main text, we determine each child's health status by the outpatient spending in the first 6 months since the child is observed in the claims data. Then, we divide children into two types (i.e., sicker or healthier) by the median spending for each age (in years) and the subsidy status at the first entry to data. There are two main complications in defining the patient health status by using the initial spending. First, each child shows up in the claims data at different ages. Second, the subsidy status may change during these months. To avoid the second issue, we focus on the individuals whose subsidy status does not change during the spell. For the first issue, we calculate the average spending separately for each age and subsidy status combination (10 years of ages groups×2 subsidy status). Then, we define those above the median of corresponding age and subsidy status as "sick" and those less than median as "healthy".

We also experiment with different windows (X=9, and 12) to calculate the average spending. The benefit of taking longer spell is that we may be able to capture the health status with more accuracy while the cost is that we may lose more observations as we impose the restriction that the subsidy status does not change during the spell (at X=6, we still maintain 90% of the total observations). Figure K-1 shows that the arc-elasticities are qualitatively similar across different *X*s.





Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month. We determine each child's health status I first X months (X= 6, 9 and 12) since one is observed in the claim data. Then, we divide children into two types (i.e., sicker or health each age (in years) and the subsidy status. The standard errors clustered at municipality level are used for estimates, and the bootstraj municipality with 200 repetitions are used for the arc-elasticities. The observations within two months from subsidy changes are exclusion anticipatory utilization. Note that the scales of *y*-axis on the arc-elasticities are set the same so that they are visually comparable.

89

(ii) Outpatient spending (in 1K JPY) **X=9**

X=6





Notes: The outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). We determine each child's health status by the outpatient spending in the first X months (X= 6, 9 and 12) since one is observed in the claim data. Then, we divide children into two types (i.e., sicker or healthier) by the median spending for each age (in years) and the subsidy status. The standard errors clustered at municipality level are used for estimates, and the bootstrapped standard errors clustered at municipality with 200 repetitions are used for the arc-elasticities. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Note that the scales of *y*-axis on the arc-elasticities are set the same so that they are visually comparable.

Appendix L: Inter-municipality migration

In the main text, we focus on the children who do not move across municipalities as there are only 1,079 such children which account for only 1.7% of total children (63,530 vs. 64,609). The migration rate in our sample is lower than the actual migration since intra-municipality migration is not counted as migration since the subsidy level is the same. However, if a family with very sick children is more likely to move to more generous municipality, our estimates —which may fail to control for the time-varying unobserved health conditions—can be potentially biased. We think that this is very unlikely for a couple of reasons. First, the migration rate is declining function of age of children and is already low by age 7 as parents tend to move before their children enter primary school. Second, there may be many other municipality characteristics than subsidy generosity for child health care that may affect the decision of migration such as other childrearing support, availability of daycare, and quality of school in the districts. Nonetheless, we include those who move across municipalities into the sample and estimate the equation [4]. Figure K-1 compares the estimates with and without movers. It is reassuring that estimates are very similar.

More direct way to test *selective* migration is to examine 1) whether children who moves are more likely to choose more generous municipality, and 2) whether sicker children are more likely to move to more generous municipality. To investigate such possibilities, we estimate a location choice model, limiting our sample to a month when children move across municipalities. Specifically, for the first question, we estimate the following equation of the conditional logit model:

$$Pr(Y_{iat} = m) = F(\sum_{A=7}^{14} \beta_A \{subsidized_{amt} \times 1(Age A)\} + \delta_m + \varepsilon_{iat}) - [L1]$$

where $Pr(Y_{iat} = m)$ is the locational choice of municipality m among M municipalities by a child i whose age is a in time t, and subsidized_{amt} is a dummy which takes one if the municipality m provides subsidy for age a in time t. We also control for municipality of choice fixed effects δ_m to control for time-invariant municipality characteristics that may attract families of the children. Our coefficients of interest are series of β_A (A=7-14) where $\beta_A > 0$ indicates that children are more likely to choose the municipality which provides the subsidy for his age a in time t. The standard errors are clustered at individual level.

For the second question, we further interact the series of subsidy dummies with the proxy for health status the average outpatient spending for six months just before the month of move (denoted by *prior spending*_{iat-1} below)²:

$$Pr(Y_{iat} = m) = F(\sum_{A=7}^{14} \beta_A \{subsidized_{amt} \times 1(Age A)\} + \sum_{A=7}^{14} \gamma_A \{subsidized_{amt} \times 1(Age A) \times prior \ spending_{iat-1}\} + \delta_m + \varepsilon_{iat}) - [L2]$$

where $\gamma_A > 0$ indicates that the sickly children are more likely to choose the municipality with subsidy. Note that in both regressions, only the children who moved across the 165 municipalities we examine in this paper are included in the sample.

The graph on the left in Figure K-2 demonstrates the graphical presentation of estimating equation [L1] which plots β_A for each age (A=7-14). Even though the estimates are quite noisy, β_A are mostly negative and are not

² We experiment the length of prior months to calculate the average prior spending from X months (X= 3,6,9 and 12) but the estimates are very similar. The benefit of taking longer span to compute the average spending is that we may be able to capture the health status with more accuracy while the cost is that we lose individual who move within the first X months from the start of the data.

statistically significant at the conventional level (See Table K-2 for the estimates). Thus, these results at least do not support that children are more likely to choose the municipality with the subsidy for her/his age.

The graph on the right in Figure K-2 demonstrates the graphical presentation of estimating equation [L2] which plots β_A in the upper half, and γ_A in the lower half. Again, β_A are not statistically significant and mostly negative. Furthermore, γ_A are close to zero, and far from statistically significant, suggesting that sickly children are no more likely to choose the municipality with subsidy. Taken together, we do not find any evidence of selective inter-municipality migration at least in the current setting.



Figure L-1: Baseline vs. including movers

Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). The dotted lines are the 95th confidence intervals derived from standard errors clustered at individual level. The observations within two months from subsidy changes are excluded from the sample to account for anticipatory utilization. Along with our baseline estimates, we report estimates from the sample that include inter-municipality movers (1.7%).



Figure L-2: Selective inter-municipality migration

Notes: The estimates from conditional logit model are plotted. The dotted lines are the 95th confidence intervals derived from standard errors clustered at individual level. The graph on the left plots β_A from estimating equation [L1] while the graphs on the right plot β_A in the upper half, and γ_A in the lower half from estimating equation [L2].

-	A. Outpatient visit dummy				B. Outpatient spending (in 1K JPY)				
	Baseline (Col. 3 Tabl	e le 2)	Including	movers	Baseline (Col. 3 Table 2)		Including movers		
	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	Estimate	(SE)	
	(1)		(2))	(3)	(4)		
Subsidized $ imes$									
Age7	0.075*** (0	0.006)	0.074^{***}	(0.005)	1.376***	(0.236)	1.351***	(0.196)	
Age8	0.068*** (0	0.006)	0.067^{***}	(0.005)	1.278***	(0.179)	1.257***	(0.181)	
Age9	0.070*** (0	0.005)	0.069***	(0.005)	1.131***	(0.171)	1.092***	(0.174)	
Age10	0.073*** (0	0.005)	0.072***	(0.005)	1.350***	(0.221)	1.340***	(0.197)	
Age11	0.066*** (0	0.006)	0.065***	(0.005)	1.113***	(0.211)	1.107^{***}	(0.197)	
Age12	0.064*** (0	0.004)	0.064***	(0.004)	0.872***	(0.223)	0.895***	(0.196)	
Age13	0.062*** (0	0.004)	0.062***	(0.004)	0.811^{***}	(0.218)	0.857^{***}	(0.216)	
Age14	0.060*** (0	0.004)	0.059***	(0.004)	0.998***	(0.134)	0.972***	(0.230)	
In-kind	0.047*** (0	0.014)	0.050***	(0.018)	0.440	(0.388)	0.532	(0.326)	
Income restriction	-0.020** (0	0.009)	-0.021***	(0.008)	-0.561	(0.372)	-0.705**	(0.298)	
R-squared	0.23		0.	.23	0.5	51	0.5	51	
Ν	2,205,64	.7	2,252	2,600	2,205,647		2,252,600		
N of Individual	63,530		64,6	509	63,5	530	64,609		
Mean wo subsidy	0.32		0.32		4.49		4.49		

Table L-1: Base vs. including movers

Notes: An outpatient visit dummy takes one if there is at least one outpatient visit per month, and outpatient spending is the monthly spending on outpatient care measured in thousand JPY (roughly USD10). For ease of comparison, Columns (1) and (3) replicate the estimates from Columns (1) and (3) in Table 2. Columns (2) and (4) report estimates from the sample that include inter-municipality movers (1.7%). The standard errors clustered at the municipality level are reported in parenthesis. *** p<0.01, ** p<0.05, * p<0.10

	No interaction with	Interaction with
	prior spending	prior spending
	(1)	(2)
Subsidized ×		
Age7	-0.352	0.281
	(0.313)	(0.439)
Age8	-0.273	-0.391
	(0.315)	(0.455)
Age9	-0.382	-0.587
	(0.298)	(0.417)
Age10	-0.150	-0.131
	(0.334)	(0.448)
Age11	-0.410	-0.675^{*}
	(0.316)	(0.369)
Age12	-0.354	-0.388
	(0.279)	(0.355)
Age13	-0.437	-0.544
	(0.285)	(0.409)
Age14	-0.055	-0.074
-	(0.315)	(0.440)
Age7 \times Prior spending		-0.070**
		(0.033)
Age8 \times Prior spending		0.008
		(0.058)
Age9 \times Prior spending		0.032
		(0.049)
Age $10 \times Prior$ spending		-0.005
		(0.050)
Age11 \times Prior spending		0.040
6 1 6		(0.027)
Age 12 \times Prior spending		-0.003
8		(0.026)
Age 13 \times Prior spending		0.054
ingene / (inter spending		(0.060)
Age 14 \times Prior spending		0.021
ingen i vinner spending		(0.058)
In-kind	2 597***	-0 891
	(0.451)	(0.755)
Income restriction	-0 484***	_0 188
meonie restriction	(0.164)	(0.318)
		(0.010)
N	161,703	126,344
N of move	1,179	920
N of Individual	1,052	816

 Table L-2: Selective inter-municipality migration

Notes: The estimates from conditional logit model are reported. Column (1) reports the results from estimating equation [L1] while column (2) results are from estimating equation [L2]. The standard errors clustered at individual level are reported in parentheses. "Prior spending" is the average outpatient spending for six months just before the month of move. The sample is limited to a month when children moves across municipalities. The number of moves is slightly larger here as the observations within two months from subsidy changes are not excluded.